UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

Form 10-Q

(Mark One)

X QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended November 30, 2017

or

TRANSITION REPORT UNDER SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from ______ to _____

Commission File Number 333-199213

ARTELO BIOSCIENCES, INC.

(Exact name of registrant as specified in its charter)

33-1220924

Nevada (State or other jurisdiction of incorporation or organization)

888 Prospect Street, Suite 210, La Jolla, CA

(IRS Employer Identification No.)

92037

(Address of principal executive offices)

(Zip Code)

<u>(760) 943-1689</u>

(Registrant's telephone number, including area code)

<u>N/A</u>

(Former name, former address and former fiscal year, if changed since last report)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. \boxtimes YES \square NO

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (\$232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). \boxtimes YES \square NO

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a small reporting company. See the definitions of "large accelerated filer", "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer Non-accelerated filer (Do not check if a smaller reporting company) Mailer reporting company Emerging growth company

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act) 🗆 YES 🛛 XO

APPLICABLE ONLY TO ISSUERS INVOLVED IN BANKRUPTCY PROCEEDINGS DURING THE PRECEDING FIVE YEARS

Check whether the registrant has filed all documents and reports required to be filed by Sections 12, 13 or 15(d) of the Exchange Act after the distribution of securities under a plan confirmed by a court. \Box YES \Box NO

APPLICABLE ONLY TO CORPORATE ISSUERS

Indicate the number of shares outstanding of each of the issuer's classes of common stock, as of the latest practicable date.

11,472,302 common shares issued and outstanding as of January 16, 2018.

TABLE OF CONTENTS

3

PART I - FINANCIAL	INFORMATION

<u>Item 1.</u>	Financial Statements	3
<u>Item 2.</u>	Management's Discussion and Analysis of Financial Condition or Plan of Operation	9
<u>Item 3.</u>	Quantitative and Qualitative Disclosures About Market Risk	15
Item 4.	Controls and Procedures	15
PART II - O	OTHER INFORMATION	16
<u>Item 1.</u>	Legal Proceedings	16
Item 1A	<u>. Risk Factors</u>	16
<u>Item 2.</u>	Unregistered Sales of Equity Securities and Use of Proceeds	18
<u>Item 3.</u>	Defaults Upon Senior Securities	18
<u>Item 4.</u>	Mine Safety Disclosures	18
<u>Item 5.</u>	Other Information	18
<u>Item 6.</u>	Exhibits	18
SIGNATU	<u>RES</u>	19

PART I - FINANCIAL INFORMATION

Item 1. Financial Statements

ARTELO BIOSCIENCES, INC. Consolidated Balance Sheets (Unaudited)

	ember 30, 2017	A	ugust 31, 2017
ASSETS			
Current Assets			
Cash and cash equivalents	\$ 412,440	\$	572,775
Prepaid expenses and deposits	16,285		1,500
Other receivable	767		-
Total Current Assets	 429,492		574,275
Equipment, net of accumulated depreciation of \$72 and \$nil, respectively	795		-
TOTAL ASSETS	 430,287		574,275
LIABILITIES AND STOCKHOLDERS' EQUITY			
Current Liabilities			
Accounts payable and accrued liabilities	\$ 133,973	\$	28,576
Due to related party	 2,308		862
Total Current Liabilities	136,281		29,438
STOCKHOLDERS' EQUITY			

Preferred Stock, par value \$0.001, 50,000,000 shares authorized, 0 and 0 shares issued and outstanding as of		
November 30, 2017, and August 31, 2017, respectively	-	-
Common Stock, par value \$0.001, 150,000,000 shares authorized, 11,352,302 and 11,327,302 shares issued and		
outstanding as of November 30, 2017 and August 31, 2017, respectively	11,352	11,327
Additional paid-in capital	855,168	827,942
Accumulated deficit	(572,146)	(295,089)
Accumulated other comprehensive gain (loss)	(368)	657
Total Stockholders' Equity	294,006	544,837
TOTAL LIABILITIES AND STOCKHOLDERS' EQUITY	\$ 430,287	\$ 574,275

The accompanying notes are an integral part of these unaudited financial statements.

ARTELO BIOSCIENCES, INC. Consolidated Statements of Operations (Unaudited)

	_	Three months ended November 30,		
		2017		2016
OPERATING EXPENSES				
General and administrative	\$	136,564	\$	313
Professional fees		107,345		9,204
Research and development		33,076		-

Depreciation		72	-
Total Operating Expenses		277,057	 9,517
Loss from Operations		(277,057)	(9,517)
NET LOSS	\$	(277,057)	\$ (9,517)
OTHER COMPREHENSIVE LOSS			
Foreign currency translation adjustments		(1,025)	 -
Total Other Comprehensive Loss		(1,025)	 -
TOTAL COMPREHENSIVE LOSS	<u>\$</u>	(278,082)	\$ (9,517)
Basic and Diluted Loss per Common Share	<u>\$</u>	(0.02)	\$ (0.00)
Basic and Diluted Weighted Average Common Shares Outstanding		11,345,635	7,640,000

The accompanying notes are an integral part of these unaudited financial statements.

ARTELO BIOSCIENCES, INC. Consolidated Statements of Cash Flows (Unaudited)

		Three months ended November 30,		
	_	2017		2016
CASH FLOWS FROM OPERATING ACTIVITIES				
Net loss	\$	(277,057)	\$	(9,517)
Depreciation		72		-
Stock based compensation		17,251		-
Changes in operating assets and liabilities:				
Prepaid expenses		(14,785)		-
Other receivable		(767)		-
Accounts payable and accrued liabilities		105,397		(2,889)
Net cash used in operating activities		(169,889)		(12,406)
CASH FLOWS FROM INVESTING ACTIVITIES				
Purchase of equipment		(867)		-

Net cash used in investing activities	(867)		-
CASH FLOWS FROM FINANCING ACTIVITIES			
Collection from stock subscription	10,000		-
Advance from related party	9,951	12,	,406
Repayment to related party	(8,505)		-
Net cash provided by financing activities	11,446	12,	,406
Effects on changes in foreign exchange rate	(1,025)		-
Net decrease in cash and cash equivalents	(160,335)		-
Cash and cash equivalents - beginning of period	572,775	3,	,590
Cash and cash equivalents - end of period	\$ 412,440	\$ 3,	,590
Supplemental Cash Flow			
Cash paid for interest	\$ 	\$	-
Cash paid for income taxes	\$ -	\$	_
Non-cash financing and investing activities:			
Loan forgiven by previous shareholder	\$ -	<u>\$</u> 16,	,856

The accompanying notes are an integral part of these unaudited financial statements.

ARTELO BIOSCIENCES, INC. Notes to the Unaudited Consolidated Financial Statements For the Three Months Ended November 30, 2017

NOTE 1 - ORGANIZATION AND DESCRIPTION OF BUSINESS

ARTELO BIOSCIENCES, INC. (the "Company") is a Nevada corporation incorporated on May 2, 2011. It is based in San Diego County, California. The accounting and reporting policies of the Company conform to accounting principles generally accepted in the United States of America, and the Company's fiscal year end is August 31.

Effective on February 10, 2017, the Company changed its name from "KNIGHT KNOX DEVELOPMENT CORP.," to "REACTIVE MEDICAL INC." On April 14, 2017, the Company changed its name from "REACTIVE MEDICAL INC." to "ARTELO BIOSCIENCES, INC".

In May 2017, the Company registered fully owned subsidiaries in England and Wales, Trinity Reliant Ventures Limited, and Trinity Research & Development Limited. Operations in the subsidiary have been consolidated in the financial statements.

The Company intends to license, develop and commercialize novel cannabinoid therapeutic treatments. To date, the Company's activities have been limited to its formation and the raising of equity capital.

NOTE 2 - SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Basis of Presentation

The Company prepares its financial statements in accordance with rules and regulations of the Securities and Exchange Commission ("SEC") and

accounting principles generally accepted ("GAAP") in the United States of America. The accompanying interim financial statements have been prepared in accordance with generally accepted accounting principles for interim financial information in accordance with Article 8 of Regulation S-X. Accordingly, they do not include all of the information and footnotes required by generally accepted accounting principles for complete financial statements. In the Company's opinion, all adjustments (consisting of normal recurring accruals) considered necessary for a fair presentation have been included. Operating results for the three months ended November 30, 2017 are not necessarily indicative of the results for the full year. While management of the Company believes that the disclosures presented herein are adequate and not misleading, these interim financial statements should be read in conjunction with the audited financial statements and the footnotes thereto for the year ended August 31, 2017 contained in the Company's Form 10-K filed on November 29, 2017.

Basis of Consolidation

The financial statements have been prepared on a consolidated basis, with the Company's fully owned subsidiary Trinity Reliant Ventures Limited. No intercompany balances or transactions exist during the period ended November 30, 2017.

NOTE 3 - GOING CONCERN

The Company's financial statements are prepared using accounting principles generally accepted in the United States of America applicable to a going concern which contemplates the realization of assets and liquidation of liabilities in the normal course of business. The Company has not established an ongoing source of revenues sufficient to cover its operating cost, and requires additional capital to commence its operating plan. The ability of the Company to continue as a going concern is dependent on the Company obtaining adequate capital to fund operating losses until it becomes profitable. If the Company is unable to obtain adequate capital, it could be forced to cease operations. These factors raise substantial doubt about its ability to continue as a going concern.

In order to continue as a going concern, the Company will need, among other things, additional capital resources. Management's plan to obtain such resources for the Company include: sales of equity instruments; traditional financing, such as loans; and obtaining capital from management and significant stockholders sufficient to meet its minimal operating expenses. However, management cannot provide any assurance that the Company will be successful in accomplishing any of its plans.

There is no assurance that the Company will be able to obtain sufficient additional funds when needed or that such funds, if available, will be obtainable on terms satisfactory to the Company. In addition, profitability will ultimately depend upon the level of revenues received from business operations. However, there is no assurance that the Company will attain profitability. The accompanying financial statements do not include any adjustments that might be necessary if the Company is unable to continue as a going concern.

The accompanying financial statements have been prepared assuming that the Company will continue as a going concern, which contemplates the realization of assets and the liquidation of liabilities in the normal course of business. During the three months ended November 30, 2017, the Company has a net loss of \$277,057. As at November 30, 2017, the Company had an accumulated deficit of \$572,146 and has earned no revenues. The Company intends to fund operations through equity financing arrangements, which may be insufficient to fund its capital expenditures, working capital and other cash requirements for future periods.

NOTE 4 - RELATED PARTY TRANSACTIONS

During the three months ended November 30, 2017, the president of the Company incurred \$440 of expenses on behalf of the Company. The amount owing to the related party as of November 30, 2017 and August 31, 2017 is \$1,302 and \$862, respectively. The amounts are non-interest bearing, and have no terms of repayment.

During the three months ended November 30, 2017, the former President, and current Senior Vice President, European Operations, who is a major shareholder paid rent expense on behalf of the Company, and paid for expenses on behalf of the company for a total of \$9,511. The amount of \$8,505 was repaid during the three months ended November 30, 2017. The amount owing to the related party as of November 30, 2017 and August 31, 2017 is \$1,006 and \$0, respectively. The amounts are non-interest bearing, and have no terms of repayment.

The Company has an employment contract with a key employee, Mr. Gregory Gorgas, who is an officer of the Company. As of November 30, 2017, no salary is owed nor has been paid.

The amounts and terms of the above transactions may not necessarily be indicative of the amounts and terms that would have been incurred had comparable transactions been entered into with independent third parties.

During the three months ended November 30, 2017, the company recorded \$17,251 of stock compensation expense for five members of the Company's Board of Directors.

NOTE 5 - EQUITY

Preferred shares

The Company has authorized 50,000,000 shares of preferred stock with a par value of \$0.001.

During the three months ended November 30, 2017, there were no issuances of preferred stock.

Common Shares

The Company has authorized 150,000,000 common shares with a par value of \$0.001 per share. Each common share entitles the holder to one vote, in person or proxy, on any matter on which action of the stockholders of the company is sought.

During the three months ended November 30, 2017, the Company received \$10,000 that has been recorded as stock issued in relation to a subscription agreement on June 30, 2017, for the issuance of 25,000 common shares.

Per the terms of the subscription agreement, following the closing date until the earlier of (i) the date that the registration is declared effective by the SEC, or (ii) the date the shares become freely tradable, if the Company issues any common stock or common stock equivalent entitling the holder to acquire common stock at a price below \$0.40, the Company will be required to issue the subscribers that number of additional unites equal to the difference between the units issued at closing, and the number units the Company would have issued to the subscriber had the offering been completed at this discounted price.

Warrants

In relation to the common stock related to subscription agreement, each individual investor received warrants with the purchase of the stock. For each share purchased, the investor will receive one Series A Common Stock Purchase Warrant to purchase one share of the Company's common stock for a period of five years from the date of the share subscription at June 30, 2017 at a price of \$1.00 per share.

As of November 30, 2017, there are 1,952,303 Series A Common Stock Purchase Warrants outstanding, with a weighted average life remaining of 4.58 years, and average exercise price of \$1.00.

NOTE 7 – SUBSEQUENT EVENTS

On December 20, 2017, the Company entered into a Material and Data Transfer, Option and License Agreement (the "License Agreement") with NEOMED Institute, a Canadian not-for-profit corporation ("NEOMED"), that provides the Company with up to twelvemonths from the date of receipt by

the Company of the required materials to conduct certain non-clinical research studies, diligence and technical analyses with NEOMED's proprietary therapeutic compound NEO1940 (the "Compound" and an option (the "Option") for an exclusive worldwide license to develop and commercialize products comprising or containing the Compound. In clinical development studies with NEOMED's prior sponsor, NEO1940 was dosed in over 200 subjects. The License Agreement has an effective date of January 2, 2018 (the "Effective Date").

NEOMED, without additional consideration and at NEOMED's sole cost, has agreed to deliver to the Company certain technology transfer materials and the quantity of the Compound substance specified in a research plan, both as set out under the License Agreement.

The Company will evaluate the Compound and then decide whether to exercise the Option. Upon exercise of the Option, NEOMED will provide the Company with an exclusive worldwide license under all of NEOMED's intellectual property rights covering the Compound ("Licensed IP Rights") to research, develop, make, have made, use, offer for sale, sell, have sold and import products containing the Compound and otherwise exploit the Licensed IP Rights in all fields.

On the Effective Date, the Company issued 120,000 shares of its common stock to NEOMED.

Item 2. Management's Discussion and Analysis of Financial Condition or Plan of Operation

This quarterly report contains forward-looking statements. These statements relate to future events or our future financial performance. In some cases, you can identify forward-looking statements by terminology such as "may", "should", "expects", "plans", "anticipates", "believes", "estimates", "predicts", "potential" or "continue" or the negative of these terms or other comparable terminology. These statements are only predictions and involve known and unknown risks, uncertainties and other factors that may cause our or our industry's actual results, levels of activity, performance or achievements to be materially different from any future results, levels of activity, performance or achievements. Although we believe that the expectations reflected in the forward-looking statements are reasonable, we cannot guarantee future results, levels of activity, performance or achievements. Except as required by applicable law, including the securities laws of the United States, we do not intend to update any of the forward-looking statements to actual results.

Our unaudited financial statements are stated in United States Dollars (US\$) and are prepared in accordance with United States Generally Accepted Accounting Principles. The following discussion should be read in conjunction with our financial statements and the related notes that appear elsewhere in this quarterly report. The following discussion contains forward-looking statements that reflect our plans, estimates and beliefs. Our actual results could

differ materially from those discussed in the forward-looking statements. Factors that could cause or contribute to such differences include, but are not limited to, those discussed below and elsewhere in this quarterly report.

Our financial statements are stated in United States Dollars (US\$) and are prepared in accordance with United States Generally Accepted Accounting Principles (GAAP).

In this quarterly report, unless otherwise specified, all dollar amounts are expressed in United States dollars and all references to "common shares" refer to the common shares in our capital stock.

As used in this quarterly report, the terms "we", "us", "our" and "our company" mean Artelo Biosciences, Inc., and our wholly owned subsidiaries, Trinity Reliant Ventures Limited, in Ireland, and Trinity Research & Development Limited, in England and Wales unless otherwise indicated.

Overview

We were incorporated under the laws of the State of Nevada on May 2, 2011 under the name Knight Knox Development Corp. Our principal address is 888 Prospect Street, Suite 210, La Jolla, California, USA and our European office is located at 29 Fitzwilliam Street, Upper, Dublin 2 Ireland. Our telephone number in North America is 760-943-1689 and our European office number is +353 (1) 443 4604.

From inception to January 2017 our business plan was that of a development stage e-commerce company with the intention of operating a fully functional auction site where customers would register for an account and sell and purchase goods and services. Beginning in April 2017, we changed our business plan and we are now focused on becoming a specialty biopharmaceutical company that intends to license, develop and commercialize novel cannabinoid therapeutic treatments, although we have licensed one provisional patent pertaining to a novel cannabinoid-based drug combination to date, we are not yet developing any such treatments.

On January 19, 2017, a majority of our stockholders and our board of directors approved a name change from Knight Knox Development Corp. to Reactive Medical Inc., to better reflect a change of direction of our business. In addition, the majority stockholder and our board of directors approved an increase to our authorized capital from 75,000,000 shares of common stock, par value \$0.001 to 150,000,000 shares of common stock, par value \$0.001. The change of name became effective with the OTC Markets at the opening of trading on February 10, 2017 under the symbol "RMED".



On March 30, 2017, Mr. Peter O'Brien resigned his positions as President, Chief Executive Officer, Chief Financial Officer, Secretary and Treasurer of the company and was appointed Senior Vice President of European Operations. On April 3, 2017, Mr. Gregory Gorgas was appointed President, Chief Executive Officer, Chief Financial Officer, Secretary, Treasurer and a member of our board of directors. On that date, the Company entered into an employment contract with Mr. Gorgas, which commits the Company and Mr. Gorgas to specific rights and responsibilities, customary to industry standards. For example, upon fulfilling certain obligations, including raising capital in excess of \$5,000,000. Mr. Gorgas will then be paid an annual salary of \$250,000 and be eligible for additional compensation in the form of bonus, equity, and benefits, commensurate with industry standards. Per the terms of the employment agreement, that any investment in, or appointment to or continuing service on a board of directors or similar body of, any be terminated in accordance with the Term of Employment specified in the agreement.

Simultaneously, on April 3, 2017, Mr. Gorgas entered into a stock purchase agreement to purchase 1,760,000 common shares for a purchase price of \$1,760.

On April 14, 2017, with the approval of its board of directors and shareholders owning a majority of our company's issued and outstanding shares by written consent in lieu of a meeting, we filed a Certificate of Change with the Secretary of State of Nevada, changing our name to Artelo Biosciences, Inc., effective as of April 28, 2017. The change of name became effective on the OTC Markets on May 2, 2017 under the symbol "ARTL".

On May 2, 2017, we entered into an Exclusive Patent License Agreement with Analog Biosciences, Inc. pursuant to which we obtained an exclusive license to two provisional patent applications, and any patent issued on such patent application, related to a combination product strategy to produce a synergy with cannabidiol which was previously assigned to Analog. We have discontinued development of the two provisional patents licensed from Analog.

On May 2, 2017, we entered into an Indemnification Agreement with its newly elected directors, Ms. Connie Matsui and Mr. Steven Kelly, who were appointed to our Board of Directors on the same date.

Pursuant to the Indemnification Agreement, our company agreed to indemnify Ms. Matsui and Mr. Kelly against all expenses, liability and loss, subject to certain limitations, arising out of their respective duties with our company. The indemnification agreement provides indemnification in addition to the indemnification provided by our company's certificate of incorporation and by-laws and by applicable law. Among other things, the Indemnification Agreement expressly provides indemnification for Ms. Matsui and Mr. Kelly for expenses, liability and loss (actually or reasonably) incurred by each of

them in connection with the investigation, defense, settlement or appeal of any proceeding relating to their respective duties with our company. In addition, we have agreed to advance expenses, subject to certain limitations, incurred by Ms. Matsui and Mr. Kelly in connection with the investigation, defense, settlement or appeal of any proceeding to which they are a party or are threatened to be made a party as a result of their respective duties with our company.

On May 4, 2017, we entered into a Note Repayment Agreement with Malibu Investments Limited, pursuant to which our company agreed to repay \$31,500, representing all of the principal and accrued interest our company owed Malibu under a Senior Promissory Note dated November 18, 2016, in the principal amount of \$30,000. The note was fully repaid during the year ended August 31, 2017.

On July 31, 2017, we entered into an indemnification agreement with Douglas Blayney, MD, who was appointed to our Board of Directors on the same date.

Pursuant to the indemnification agreement, we agreed to indemnify Dr. Blayney against all expenses, liability and loss, subject to certain limitations, arising out of his respective duties with our company. The indemnification agreement provides indemnification in addition to the indemnification provided by our company's certificate of incorporation and by-laws and by applicable law. Among other things, the indemnification agreement expressly provides indemnification for Dr. Blayney for expenses, liability and loss (actually or reasonably incurred by each of them in connection with the investigation, defense, settlement or appeal of any proceeding relating to their respective duties with our company. In addition, we have agreed to advance expenses,

subject to certain limitations, incurred by Dr. Blayney in connection with the investigation, defense, settlement or appeal of any proceeding to which he is a party or are threatened to be made a party as a result of his respective duties with our company.

On August 1, 2017, Mr. Peter O'Brien, a member of our Board of Directors and our Senior Vice President – European Operations and our company entered into a stock purchase agreement with ALII Capital LLC, a Washington limited liability corporation pursuant to which Mr. O'Brien sold 300,000 shares of our stock owned by him for \$300. Pursuant to the terms of the agreement, we granted ALII Capital demand registration rights for the shares purchased.

On September 20, 2017, we entered into indemnification agreements with each of Ms. Georgia Erbez and R. Martin Emanuele, PhD, who were appointed to our Board of Directors on the same date.

Pursuant to the indemnification agreements, we agreed to indemnify Ms. Erbez and Dr. Emanuele against all expenses, liability and loss, subject to certain limitations, arising out of their respective duties with our company. The indemnification agreements provide indemnification in addition to the indemnification provided by our company's certificate of incorporation and by-laws and by applicable law. Among other things, the indemnification agreements expressly provides indemnification for Ms. Erbez and Dr. Emanuele for expenses, liability and loss (actually or reasonably incurred by each of them in connection with the investigation, defense, settlement or appeal of any proceeding relating to their respective duties with our company. In addition, we have agreed to advance expenses, subject to certain limitations, incurred by Ms. Erbez and Dr. Emanuele in connection with the investigation, defense, subject to certain limitations, incurred by Ms. Erbez and Dr. Emanuele in connection with the investigation of any proceeding to which they are a party or are threatened to be made a party as a result of their respective duties with our company.

On December 20, 2017, the Company entered into a Material and Data Transfer, Option and License Agreement (the "License Agreement") with NEOMED Institute, a Canadian not-for-profit corporation ("NEOMED"), that provides the Company with up to twelvemonths from the date of receipt by the Company of the required materials to conduct certain non-clinical research studies, diligence and technical analyses with NEOMED's proprietary therapeutic compound NEO1940 (the "Compound" and an option (the "Option") for an exclusive worldwide license to develop and commercialize products comprising or containing the Compound. In clinical development studies with NEOMED's prior sponsor, NEO1940 was dosed in over 200 subjects. The License Agreement has an effective date of January 2, 2018 (the "Effective Date").

NEOMED, without additional consideration and at NEOMED's sole cost, has agreed to deliver to the Company certain technology transfer materials and the quantity of the Compound substance specified in a research plan, both as set out under the License Agreement.

The Company will evaluate the Compound and then decide whether to exercise the Option. Upon exercise of the Option, NEOMED will provide the Company with an exclusive worldwide license under all of NEOMED's intellectual property rights covering the Compound ("Licensed IP Rights") to research, develop, make, have made, use, offer for sale, sell, have sold and import products containing the Compound and otherwise exploit the Licensed IP Rights in all fields.

We are a discovery research and development stage company and have commenced only minimal business operations and have not generated any revenues. We have been issued a "going concern" opinion by our auditor, based upon our reliance on the sale of our common stock as the sole source of funds for our current operations.

We have two wholly owned subsidiaries, Trinity Reliant Ventures Limited, in Ireland, and Trinity Research & Development Limited, in England and Wales.

We have never declared bankruptcy, been in receivership, or involved in any kind of legal proceeding.

Our Current Business

We are an ethical development-stage biopharmaceutical company focused on discovering, licensing, developing and commercializing treatments that modulate the endocannabinoid system. We intend to pursue technologies and programs that offer promising and proprietary approaches to cannabinoid-based therapies, including those derived from the cannabis plant and synthetic cannabinoids, as well as new chemical entities and compounds that promote the effectiveness of the endocannabinoid system. We are currently evaluating programs in each of the following areas: phytocannabinoids, synthetics, and new chemical entities. Our flagship program is designed to be a patent-protected cannabinoid drug combination treatment for a rare and orphan disease with vital unmet medical needs. We believe our programs have the potential to dramatically improve patient care in major markets, including the United States and Europe.

Our board and management have experience developing and commercializing ethical pharmaceutical products, including several first-in-class drugs in multiple therapeutic areas. As we build our pipeline and advance programs through the research and development process, we expect to evaluate partnerships with large pharmaceutical and biopharmaceutical companies to collaborate on research, support clinical development, and enter into commercial licensing agreements. We intend to preserve our development and commercialization rights while embracing collaborations without hesitation in certain situations and territories where we believe there is a strong driver for maximum value creation.

To date, none of our product candidates have completed clinical development, been submitted for regulatory review or received marketing authorization from any regulatory agency. Therefore, we have not yet received revenue from the sale of any of our product candidates.

Results of Operations

The following summary of our results of operations, for the three months ended November 30, 2017 and 2016, should be read in conjunction with our interim financial statements, as included in this Form 10-Q and our audited financial statements for the year ended August 31, 2017, as included in Form 10-K filed with the SEC on November 29, 2017.

Our financial statements have been prepared assuming that we will continue as a going concern and, accordingly, do not include adjustments relating to the recoverability and realization of assets and classification of liabilities that might be necessary should we be unable to continue in operation. We expect we will require additional capital to meet our long term operating requirements. We expect to raise additional capital through, among other things, the sale

of equity or debt securities, but we cannot guarantee that we will be able to achieve same.

The following table provides selected financial data about our company as of November 30, 2017 and August 31, 2017.

Balance Sheet Data

	Novembe 2017	November 30, Aug 2017 2	
Cash	\$ 41	2,440	\$ 572,775
Total Assets	\$ 43	,287	\$ 574,275
Total Liabilities	\$ 13	,281	\$ 29,438
Stockholders' Equity	\$ 29	,006	\$ 544,837

We have not generated any revenues since inception through November 30, 2017. The decrease in cash was primarily due to payments of expenses.

For the Three Months Ended November 30, 2017 Compared to the Three Months Ended November 30, 2016

	Three Month Novembe			
	 2017		2016	
Operating Expenses				
General and administrative expense	\$ 119,313	\$	313	
Stock based compensation	17,251		-	
Professional fees	107,345		9,204	
Research and development	33,076		-	
Depreciation	72		-	
Total Operating Expenses	277,057		9,517	
Loss from Operations	(277,057)		(9,517)	
Provision for income taxes	_		-	
Net Loss	\$ (277,057)	\$	(9,517)	

Our operating expenses, for the three months ended November 30, 2017 were \$277,057 compared to \$9,517 for the same period in 2016. The increase in operating expenses were primarily due to the increased activity in developing our proposed business plan, which we expect will continue for the foreseeable future.

Liquidity and Capital Resources

Working Capital

	November 30, 2017	August 31, 2017
Current Assets	\$ 429,492	\$ 574,275
Current Liabilities	136,281	29,438
Working Capital	\$ 293,211	\$ 544,837

Cash Flows

	Three months Ended November 30,	
	2017	2016
Cash Flows used in operating activities	\$ (169,899)	\$ (12,406)
Cash Flows used in investing activities	(867)	-
Cash Flows provided by financing activities	11,446	12,406
Net decrease in cash during period	\$ (160,335)	\$-

Cash Flow from Operating Activities

During the three months ended November 30, 2017, cash used in operating activities was \$169,889 compared to cash used in operating activities of \$12,406 during the period ended November 30, 2016. The cash used from operating activities during the three months ended November 30, 2017 primarily consisted of a net loss of \$277,057 and a decrease in prepaid of \$14,785, offset by stock based compensation of \$17,251 and an increase in accounts payable and accrued liabilities of \$14,785. The cash used from operating activities during the three months ended November 30, 2016 consisted of a net loss of \$9,517 and a decrease in accounts payable and accrued liabilities of \$2,889.

Cash Flow from Investing Activities

The company used \$867 for a purchase of equipment during the three months ended November 30, 2017 The company did not use any funds for investing activities during the three months ended November 30, 2016.

Cash Flow from Financing Activities

During the three months ended November 30, 2017, the company received \$9,951 as an advance from a related party, \$10,000 the issuance of common shares and repaid \$8,505 to a related party. During the three months ended November 30, 2016, the company received \$12,406 as an advance from a related party.

Going Concern

Our auditors issued a going concern opinion on our financial statements as of and for the period ended August 31, 2017. This means that there is

substantial doubt that we can continue as an on-going business for the next twelve months unless we obtain additional capital to pay for our expenses. This is because we have not generated sufficient revenues to cover operating costs or raised enough funds. There is no assurance we will ever reach this point. Accordingly, we must raise sufficient capital from sources. We must raise cash to stay in business. In response to these problems, management intends to raise additional funds through public or private placement offerings. At this time, however, the Company does not have plans or intentions to raise additional funds by way of the sale of additional securities, other than pursuant to our current Offering.

Off Balance Sheet Arrangement

We do not have any relationships with unconsolidated entities or financial partnerships, such as entities often referred to as structured finance or special purpose entities, which would have been established for the purpose of facilitating off-balance sheet arrangements or other contractually narrow or limited purposes. In addition, we do not have any undisclosed borrowings or debt, and we have not entered into any synthetic leases. We are, therefore, not materially exposed to any financing, liquidity, market, or credit risk that could arise if we had engaged in such relationships.

Critical Accounting Policies and Estimates

We prepare our financial statements in conformity with GAAP, which requires management to make certain estimates and apply judgments. We base our estimates and judgments on historical experience, current trends and other factors that management believes to be important at the time the financial statements are prepared. On a regular basis, we review our accounting policies and how they are applied and disclosed in our financial statements.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

As a "smaller reporting company", we are not required to provide the information required by this Item.

Item 4. Controls and Procedures

Management's Report on Disclosure Controls and Procedures

As of November 30, 2017, management assessed the effectiveness of our internal control over financial reporting based on the criteria for effective internal control over financial reporting established in Internal Control-Integrated Framework issued by the Committee of Sponsoring Organizations of the Treadway Commission ("**COSO**") in Internal Control-Integrated Framework (2013), and SEC guidance on conducting such assessments. Based on that evaluation, they concluded that, during the period covered by this report, such internal controls and procedures were not effective to detect the inappropriate application of US GAAP rules as more fully described below. This was due to deficiencies that existed in the design or operation of our internal controls over financial reporting that adversely affected our internal controls and that may be considered to be material weaknesses.

The matters involving internal controls and procedures that our management considered to be material weaknesses under the standards of the Public Company Accounting Oversight Board were: (1) lack of a functioning audit committee; (2) inadequate segregation of duties consistent with control objectives; and (3) management dominated by a single individual without adequate compensating controls. The aforementioned material weaknesses were identified by our Chief Executive and Financial Officer in connection with the review of our financial statements as of November 30, 2017.

Management believes that the material weaknesses set forth above did not have an effect on our financial results. However, management believes that the lack of a functioning audit committee and the lack of a majority of outside directors on our board of directors results in ineffective oversight in the establishment and monitoring of required internal controls and procedures, which could result in a material misstatement in our financial statements in future periods.

Changes in Internal Control Over Financial Reporting

During the period covered by this report there were no changes in our internal control over financial reporting that materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II - OTHER INFORMATION

Item 1. Legal Proceedings

From time to time, we may become involved in litigation relating to claims arising out of its operations in the normal course of business. We are not involved in any pending legal proceeding or litigation and, to the best of our knowledge, no governmental authority is contemplating any proceeding to which we area party or to which any of our properties is subject, which would reasonably be likely to have a material adverse effect on us.

Item 1A. Risk Factors

Investing in our common stock involves risk. Before making an investment in our common stock, you should carefully consider the risk factors discussed in Part I, Item 1A, "Risk Factors" of our Form 10-K for the year ended August 31, 2017. The risks described in the Form 10-K are those which we believe are the material risks we face, and such risks could materially adversely affect our business, prospects, financial condition, cash flows and results of operations. Additional risks and uncertainties not currently known to us or that we currently deem to be immaterial also may impact us. Except as set forth below, there have been no material changes in our risk factors from those previously disclosed in the Form 10-K.

If we fail to comply with our obligations to our licensor in our intellectual property license, we could lose license rights that are important to our business.

We are a party to the NEOMED Agreement, and we may enter into additional license agreements in the future. Our existing license agreements impose, and we expect that any future license agreements will impose, various diligence, product payment, royalty, insurance and other obligations on us. If we fail to comply with these obligations, our licensors may have the right to terminate these agreements, in which event we might not be able to develop and market any product candidate that is covered by these agreements. Termination of these licenses or reduction or elimination of our licensed rights may result in our having to negotiate new or reinstated licenses with less favorable terms. The occurrence of such events could have a material adverse effect on our business, financial condition and results of operations.

Even if we are successful in licensing or developing research programs and/or product candidates, we or our licensors must maintain the intellectual property.

Our commercial success is significantly dependent on intellectual property related to any product candidates and technologies we may either acquire, license or develop internally. We are currently the licensee of two patent applications; however we intend to license additional technologies from pharmaceutical and biotechnology companies, and research institutions. In addition, based upon our own discovery research initiatives, we filed a provisional patent application on December 11, 2017 on novel chemistry related to a potential cannabinoid formulation. We have not received action on any of the provisional applications whether obtained as licenses or as a result of our own research efforts.

Our success depends in large part on our and our licensor's ability to obtain and maintain patent protection in the United States and other countries with respect to our proprietary technology and product candidates. In some circumstances, we may not have the right to control the preparation, filing and prosecution of patent applications, or to maintain the patents, covering technology or products that we license from third parties. Therefore, we cannot be certain that these patents and applications will be prosecuted and enforced in a manner consistent with the best interests of our business. In addition, if third parties who license patents to us fail to maintain such patents, or lose rights to those patents, the rights we have licensed may be reduced or eliminated.

The patent position of biotechnology and pharmaceutical companies generally is highly uncertain, involves complex legal and factual questions and has in recent years been the subject of much litigation. As a result, the issuance, scope, validity, enforceability and commercial value of our and our licensor's patent rights are highly uncertain. Our and our licensor's pending and future patent applications may not result in patents being issued which protect our technology or products or which effectively prevent others from commercializing competitive technologies and products. Changes in either the patent laws or interpretation of the patent laws in the United States and other countries may diminish the value of our patents or narrow the scope of our patent protection. The laws of foreign countries may not protect our rights to the same extent as the laws of the United States. Publications or discoveries in the scientific literature often lag behind the actual discoveries, and patent applications in the United States and other jurisdictions are typically not published until 18 months after filing, or in some cases not at all. Therefore, we cannot be certain that we or our licensor were the first to make the inventions claimed in our owned and licensed patents for patentability are met, the first to file a patent application is entitled to the patent. We may become involved in opposition or interference proceedings challenging our patent rights or the patent rights of others. An adverse determination in any such proceeding could reduce the scope of, or invalidate our patent rights, allow third parties to commercialize our product candidates without infringing third-party patent rights.

Even if any owned and/or licensed patent applications issue as patents, they may not issue in a form that will provide us with any meaningful protection, prevent competitors from competing with us or otherwise provide us with any competitive advantage. Our competitors may be able to circumvent our owned or licensed patents by developing similar or alternative technologies or products in a non-infringing manner. The issuance of a patent is not conclusive as to its scope, validity or enforceability, and our owned and licensed patents may be challenged in the courts or patent offices in the United States and abroad. Such challenges may result in patent claims being narrowed, invalidated or held unenforceable, which could limit our ability to stop or prevent us from stopping others from using or commercializing similar or identical technology and products, or limit the duration of the patent protection of our technology and product candidates. Given the amount of time required for the development, testing and regulatory review of new product candidates, patents protecting such candidates might expire before or shortly after such candidates are commercialized. As a result, our owned and licensed patents from commercialized to ours.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

None.

Item 3. Defaults Upon Senior Securities

None.

Item 4. Mine Safety Disclosures

Not Applicable.

Item 5. Other Information

None.

Item 6. Exhibits

Exhibit	Description
Number	-
(10)	Material contracts
10.1	Material and Data Transfer, Option and License Agreement dated December 20, 2017 between the Company and NEOMED
	Institute*+
(31)	Rule 13a-14 (d)/15d-14d) Certifications
31.1*	Section 302 Certification by the Principal Executive Officer, Principal Financial Officer and Principal Accounting Officer
<u>31.1*</u> (32)	Section 1350 Certifications
<u>32.1*</u> 101	Section 906 Certification by the Principal Executive Officer, Principal Financial Officer and Principal Accounting Officer
101	Interactive Data File
101.INS**	XBRL Instance Document
101.SCH**	XBRL Taxonomy Extension Schema Document
101.CAL**	XBRL Taxonomy Extension Calculation Linkbase Document
101.DEF**	XBRL Taxonomy Extension Definition Linkbase Document
101.LAB**	XBRL Taxonomy Extension Label Linkbase Document
101.PRE**	XBRL Taxonomy Extension Presentation Linkbase Document

* Filed herewith

** Furnished herewith. XBRL (eXtensible Business Reporting Language) information is furnished and not filed or a part of a registration statement or prospectus for purposes of Sections 11 or 12 of the Securities Act of 1933, as amended, is deemed not filed for purposes of Section 18 of the Securities Exchange Act of 1934, as amended, and otherwise is not subject to liability under these sections.

+ Portions of this exhibit have been omitted pursuant to a request for confidential treatment and the non-public information has been filed separately with the SEC.

SIGNATURES

In accordance with Section 13 or 15(d) of the Exchange Act, the registrant caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

 Dated: January 16, 2018
 ARTELO BIOSCIENCES, INC. (Registrant)

 January 16, 2018
 /s/ Greg Gorgas

 Greg Gorgas
 Greg Gorgas

 President, Chief Executive Officer, Chief Financial Officer, Treasurer and Director (Principal Executive Officer, Principal Financial Officer and Principal Accounting Officer)

CONFIDENTIAL TREATMENT REQUESTED

CONFIDENTIAL PORTIONS OF THIS DOCUMENT HAVE BEEN REDACTED AND HAVE BEEN SEPARATELY FILED WITH THE SECURITIES AND EXCHANGE COMMISSION. INFORMATION THAT WAS OMITTED IN THE EDGAR VERSION HAS BEEN NOTED IN THIS DOCUMENT WITH A PLACEHOLDER IDENTIFIED BY THE MARK "[***]".

MATERIAL AND DATA TRANSFER, OPTION AND LICENSE AGREEMENT

THIS MATERIAL AND DATA TRANSFER, OPTION AND LICENSE AGREEMENT (this " Agreement") entered into on this December 20, 2017 (the "Signature Date") by and between NEOMED Institute, a not-for-profit corporation established under the *Not-for-Profit Corporations Act* (Canada), having an address at 7171 Frederick-Banting, Saint-Laurent, Quebec H4S 1Z9, Canada ("NEOMED"), and Artelo Biosciences, Inc., a Nevada corporation, having an address at 888 Prospect Street, Suite 210, La Jolla, California 92037, U.S.A. ("Artelo") shall be effective as of the Effective Date (as defined in Section 1.9 below). Each of NEOMED and Artelo may be referred to herein as a "Party", or jointly as the "Parties".

WHEREAS, NEOMED owns or controls rights in and to its proprietary therapeutic compound NEO1940 (formally known as [***]);

WHEREAS, Artelo desires to obtain from NEOMED an exclusive option for an exclusive worldwide license to develop and commercialize products comprising or containing the NEO1940 compound;

WHEREAS, Artelo desires to obtain from NEOMED certain materials, data and information relating to the NEO1940 compound for evaluation, research and development use in order to assist Artelo in making the determination whether to exercise such exclusive option; and

WHEREAS, the Parties agree that in the event that Artelo exercises such exclusive option, NEOMED shall grant to Artelo such exclusive worldwide license, subject to and in accordance with the terms and conditions of this Agreement;

NOW, THEREFORE, in consideration of the foregoing premises and the mutual covenants herein contained, the Parties agree as follows:

1. DEFINITIONS

1.1 "Affiliate" shall mean, with respect to any Person, any other Person which directly or indirectly controls, is controlled by, or is under common control with, such Person. A Person shall be regarded as in control of another Person if it owns, or directly or indirectly controls, more than fifty percent (50%) of the voting stock or other ownership interest of the other Person, or if it directly or indirectly possesses the power to direct or cause the direction of the management and policies of the other Person by any means whatsoever.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

1.2 "**Commercially Reasonable Efforts**" shall mean, with respect to the efforts and resources to be expended by a Party, efforts and resources commensurate with the efforts and resources commonly used in the pharmaceutical or biotechnology industry by a company of comparable size in connection with the development or commercialization of pharmaceutical or biotechnology products that are of similar status, taking into account the proprietary position of the product (including intellectual property scope, subject matter and coverage), safety and efficacy, product profile, competitiveness of the marketplace, the regulatory status and approval process, anticipated or approved labeling, present and future market potential, the probable profitability of the applicable product (including pricing and reimbursement status achieved or likely to be achieved) and other relevant factors such as technical, legal, scientific or medical factors.

1.3 "Compound" shall mean NEOMED's proprietary therapeutic compound NEO1940 (formerly known as [***]).

1.4 "Compound-Related Improvements" shall mean any and all IP Improvements relating solely to the Compound.

1.5 "Confidential Information" shall mean, with respect to a Party ('Discloser"), all information of any kind whatsoever, and all tangible and intangible embodiments thereof of any kind whatsoever, which is or was disclosed by or on behalf of such Party to the other Party ("Recipient") in connection with this Agreement or the Confidentiality Agreement. Notwithstanding the foregoing, Confidential Information of a Party shall not include information which Recipient can establish by written documentation: (a) to have been publicly known prior to disclosure of such information by Discloser to Recipient, (b) to have been received by Recipient at any time from a source, other than Discloser, rightfully having possession of and the right to disclosure of such information without restriction, (d) to have been otherwise known by Recipient prior to disclosure of such information by Discloser to Recipient or (e) to have been independently developed by employees or agents of Recipient without access to or use of such information discloser to Recipient.

1.6 "Confidentiality Agreement" shall mean the Confidentiality Agreement between the Parties dated September 13, 2017.

1.7 "**Control**", "**Controls**" or "**Controlled**" shall mean, with respect to any know-how, patents, copyrights, proprietary information or trade secrets, or other intellectual property rights (collectively, "**Rights**"), the legal authority or right (whether by ownership, license or otherwise) of a Party to grant a license or a sublicense of or under such Rights to the other Party, or to otherwise disclose such proprietary information or trade secrets to the other Party, without breaching the terms of any agreement with a Third Party, or misappropriating the proprietary information or trade secrets of a Third Party.

1.8 "Data" means all data, including, and not limited to, clinical study reports, clinical study protocols, clinical data, and historical clinical safety data that is necessary or useful to seek regulatory approval to market the Product, as well as all regulatory reports, regulatory communications and other regulatory information developed, received or prepared by or for Artelo for the Product, except for any and all patient information;

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.



1.9 "Effective Date" shall mean January 2, 2018.

1.10 "FDA" shall mean the United States Food and Drug Administration or any successor agency thereto.

1.11 "FD&C Act" shall mean the United States Federal Food, Drug, and Cosmetic Act (21 U.S.C. § 301 et seq.), as amended, and the rules and regulations promulgated thereunder.

1.12 "Field" shall mean all fields of use.

1.13 **"First Commercial Sale**" shall mean, with respect to a Product in any country, the first sale of such Product in such country to a Third Party for consideration. First Commercial Sale excludes any sale or other distribution of a Product for use in a clinical trial or other development activity, reasonable promotional use (including samples) prior to marketing approval or for reasonable compassionate use or on a named patient basis.

1.14 "Indication" shall mean each separate disease, disorder, illness, health condition, or interruption, cessation or disruption of a bodily function, system, tissue type or organ and all of its associated signs, symptoms, stages or progression (including precursor conditions) ("Condition") for which a separate Regulatory Approval is required. Notwithstanding the foregoing, subpopulations or patients with a primary Condition, however stratified (including stratification by stages of progression, particular combinations of symptoms associated with the primary disease or condition, prior treatment courses, response to prior treatment, family history, clinical history, genotype, phenotype or other stratification) shall not be deemed to be separate Indications for the purposes of this Agreement. For clarity, different cancer types (breast, colon, lung, etc.) are each considered an Indication while stages of progression within a single cancer type are not considered an Indication under this definition.

1.15 "**Improvement**" shall mean, with respect to any intellectual property, any revision, modification, translation, abridgment, condensation, expansion, adaptation, addition, improvement, derivative work, update or upgrade to or from such intellectual property.

1.16 **"IND**" shall mean an Investigational New Drug Application, as defined in the FD&C Act, that is required to be filed with the FDA before beginning clinical testing of a Product in human subjects, or a comparable foreign filing.

1.17 "IP Improvement" shall mean any Improvement to any intellectual property covered by the Licensed IP Rights made by Artelo, its Affiliates or sublicensees after the date of the Option Exercise Notice.

1.18 "Know-How" shall mean all information and data that is not generally known (including, but not limited to, information and data regarding formulae, procedures, protocols, techniques, pharmacological, toxicological and clinical data and results and other results of experimentation and testing).

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 3 -

1.19 "Licensed IP Rights" shall mean, collectively, the Licensed Patent Rights, the Licensed Copyrights and the Licensed Know-How

Rights.

1.20 "Licensed Copyrights" shall mean rights in works of authorship, including without limitation copyrights, whether registered or unregistered, arising under the laws of any jurisdiction anywhere in the world, including moral rights and mask work rights, and all registrations and applications for registration with respect thereto which are Controlled by NEOMED as of the Effective Date or at any time during the term of this Agreement and which are necessary or useful for Artelo and its Affiliates and sublicensees to use, develop, sell, or seek regulatory approval to market or otherwise exploit the Compound.

1.21 "Licensed Know-How" shall mean all Know-How which is Controlled by NEOMED as of the Effective Date or at any time during the term of this Agreement and which is necessary or useful for Artelo and its Affiliates and sublicensees to use, develop, sell, or seek regulatory approval to market or otherwise exploit the Compound.

1.22 "Licensed Know-How Rights" shall mean all trade secret, know-how and other intellectual property rights (other than Licensed Copyrights and Licensed Patent Rights) in the Licensed Know-How.

1.23 "Licensed Patent Rights" shall mean (a) all patent applications heretofore or hereafter filed or having legal force in any country within the Territory which claim any discovery or inventions relating to the Compound, or the process of manufacture or use thereof Controlled by NEOMED as of the Effective Date or at any times during the term of this Agreement, including without limitation those listed on Exhibit A attached hereto; (b) all patents that have issued or in the future issue from such patent applications, including utility, model and design patents and certificates of invention Controlled by NEOMED as of the Effective Date or at any times during the term of this Agreement, including without limitation those listed on Exhibit A attached hereto; and (c) all divisionals, continuations, continuations-in-part, reissues, renewals, extensions or additions to any such patent applications and patents.

1.24 "Major Market Countries" shall mean the United States, Canada, the United Kingdom, Germany, France, Spain, Italy and Japan.

1.25 "NDA" shall mean a new drug application or product license application or its equivalent filed with and accepted by the FDA after completion of human clinical trials to obtain marketing approval for any Product, or any comparable application filed with and accepted by the Regulatory Authority of a country other than the United States.

1.26 "Net Sales" shall mean, with respect to any Product, the invoiced sales price of such Product billed by Artelo, its Affiliates and sublicensees to Third Party customers, less reasonable and customary (a) credits, allowances, discounts and rebates to, and chargebacks from the account of, such independent customers for spoiled, damaged, out-dated, rejected or returned Product; (b) prepaid actual freight and insurance costs incurred in transporting such Product to such customers separately invoiced and stated; (c) cash, quantity and trade discounts and other price reductions in amounts customary in the trade for quantity purchases; (d) sales, use, value-added and other direct taxes incurred separately invoiced and borne by Artelo, its Affiliates and sublicensees; and (e) customs duties, surcharges and other governmental charges incurred in connection with the exportation or importation of such Product separately invoiced and borne by Artelo, its Affiliates and development, clinical trials, compassionate use, humanitarian and charitable donations, indigent programs or for use as samples will be excluded from the computation of Net Sales.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 4 -

1.27 "Option" shall have the meaning set forth in Section 3.1.

1.28 "Option Exercise Notice" shall have the meaning set forth in Section 3.2.

1.29 "Option Period" shall have the meaning set forth in Section 3.1.

1.30 "**Person**" shall mean an individual, corporation, partnership, limited liability company, trust, business trust, association, joint stock company, joint venture, pool, syndicate, sole proprietorship, unincorporated organization, governmental authority or any other form of entity not specifically listed herein.

1.31 "Phase II Clinical Study" shall mean a study of a Product in human patients to determine initial efficacy and dose range finding before embarking on Phase III Clinical Studies.

1.32 "**Phase III Clinical Study**" shall mean a study in human patients with a defined dose or a set of defined doses of a Product which study, if the defined end-points are met, is designed to ascertain or establish efficacy and safety of such Product in patients for the indication being studied for the purpose of preparing and submitting a Regulatory Approval Application to the competent Regulatory Authority in a country of the world.

1.33 **"Product**" shall mean any product containing or comprising the Compound and/or containing, prepared with, using or covered by Licensed IP Rights or any part thereof.

1.34 "Product-Related Improvements" shall mean any and all IP Improvements (other than Compound-Related Improvements) that are reasonably necessary to develop, make or sell Compound or Product.

1.35 "**Regulatory Approval**" shall mean any technical, medical and scientific license, registration, authorization or approval (including, without limitation, any approval of an NDA, supplement or amendment, pre- and post- approval, pricing approval, or labeling approval) of any national, supra-national, regional, state or local regulatory agency, department, bureau, commission, council or other governmental entity, necessary for the commercial manufacture, distribution, marketing, promotion, offer for sale, use, import, export and sale of a Product in a regulatory jurisdiction.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 5 -

1.36 **"Regulatory Approval Application"** shall mean an application submitted to the appropriate Regulatory Authority seeking Regulatory Approval of a Product for use in one or more therapeutic indications in a regulatory jurisdiction within the Territory.

1.37 "**Regulatory Authority**" shall mean any national, regional, state or local regulatory agency, department, bureau, commission, council or other governmental entity in each country of the world involved in the granting of Regulatory Approval for the Product.

1.38 "Research Plan" shall mean the research plan attached hereto as Exhibit B attached hereto.

1.39 **"Royalty Term**" shall mean, with respect to each Product in each country, the period until the [***] of the expiration of the last to expire Valid Claim in such country that would be infringed, or if such Valid Claim is a pending patent application, would be infringed if such application were to issue with the claims as then being prosecuted, as applicable to such Product in such country, but for the license granted by this Agreement. For the absence of doubt, as an example, [***].

1.40 "**Technology Transfer Materials**" means all information, materials, and documentation Controlled by NEOMED relating to the Compound which is reasonably necessary or useful (a) for Artelo's evaluation, research and development use in order to assist Artelo in making the determination whether to exercise the Option, and (b) if Artelo exercises the Option, for Artelo's use to research, develop and commercialize the Products, including (i) the quantity of Compound specified in the Research Plan; and (ii) any other data, information and documents Controlled by NEOMED relating to the Compound and known to NEOMED to be necessary or reasonably useful for Artelo to exercise its rights to perform any authorized activities under this Agreement.

1.41 "Territory" shall mean the entire world.

1.42 "Third Party" shall mean any Person other than NEOMED, Artelo and their respective Affiliates.

1.43 **"Valid Claim**" shall mean either (a) a claim of an issued and unexpired patent within the Licensed Patent Rights, which has not been held permanently revoked, unenforceable or invalid by a decision of a court or other governmental agency of competent jurisdiction, unappealable or unappealed within the time allowed for appeal, and which has not been admitted to be invalid or unenforceable through reissue or disclaimer or otherwise; or (b) a claim of a pending patent application within the Licensed Patent Rights.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 6 -

2. MATERIAL AND DATA TRANSFER

2.1 Supply of Technology Transfer Materials.

2.1.1 Within [***] after the Effective Date, NEOMED, without additional consideration and at NEOMED's sole cost, shall deliver to Artelo the Technology Transfer Materials set forth on Exhibit C in its possession and such quantity of Compound substance as mutually agreed to by the parties ("**Required Compound Supply**").

2.1.2 If Artelo exercises the Option, promptly following the receipt of the Option Exercise Notice, NEOMED shall without additional consideration and at NEOMED's sole cost, deliver to Artelo, without limitation, all Technology Transfer Materials and all remaining quantities of Compound substance in NEOMED's possession. For the avoidance of doubt, NEOMED shall have no further obligation to provide Compound substance in addition to the quantity in NEOMED's possession as of the Effective Date, and if Artelo requires additional supply of the Compound substance in addition to the such quantity of Compound substance in NEOMED's possession, ARTELO will be responsible at its own cost for its sourcing, synthesis, and supply of such additional Compound substance.

2.1.3 If at any time during the term of this Agreement either Party discovers any additional Technology Transfer Materials (other than the Compound substance) which are Controlled by NEOMED but which were not provided by NEOMED to Artelo , or (ii) any updates to or new versions of any Technology Transfer Materials provided by NEOMED to Artelo pursuant to this Section 2.1 become available, NEOMED shall provide requested, updates to, or new versions of, as applicable, Technology Transfer Materials to Artelo, without additional consideration, promptly following the discovery thereof or as any such update or new version of such Technology Transfer Materials become available, as applicable.

2.2 Use of Technology Transfer Materials.

2.2.1 Unless and until Artelo exercises the Option, Artelo (a) shall use the Technology Transfer Materials solely to conduct the activities set forth in the Research Plan for evaluation, research and development purposes in order to assist Artelo in making the determination whether to exercise the Option, and (b) shall not file or prosecute in any country any patent application which claims or uses or purports to claim or use the Technology Transfer Materials or their use or anything resulting from their use.

2.2.2 Unless and until Artelo exercises the Option, promptly upon the expiration of the Option Period or earlier termination of this Agreement, Artelo shall return all remaining Technology Transfer Materials to NEOMED.

2.2.3 If Artelo's activities conducted pursuant to the Research Plan prior to the exercise of the Option by Artelo result in any invention or discovery that constitutes an Improvement to the Technology Transfer Materials, such invention or discovery and all intellectual property rights therein shall be owned solely by NEOMED. Artelo shall promptly disclose each such invention or discovery to NEOMED. Artelo hereby assigns to NEOMED all right, title and interest in all such inventions and discoveries and all intellectual property rights therein ("**Research Program IP**"). Artelo shall perform such acts and sign and cause to be signed such documents (including assignments) necessary or useful to ensure that all rights in the Research Program IP vest with NEOMED at all times. For clarity, if Artelo exercises the Option, the Research Program IP shall be included in the Licensed IP Rights.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 7 -

2.2.4 Artelo hereby acknowledges that the Technology Transfer Materials are experimental in nature and that they are provided "AS IS." NEOMED MAKES NO REPRESENTATIONS OR WARRANTIES, EXPRESS OR IMPLIED, WITH RESPECT TO THE TECHNOLOGY TRANSFER MATERIALS OR THE USE THEREOF. Artelo shall comply with all laws and governmental rules, regulations and guidelines which are applicable to the Technology Transfer Materials or the use thereof, including without limitation biosafety procedures, and with any safety precautions accompanying the Technology Transfer Materials.

3. OPTION

3.1 Option Grant. For a period commencing on the Effective Date and subject to early termination, ending on [***] NEOMED delivers the Required Compound Supply to Artelo ("Option Period"), Artelo shall have the sole and exclusive right, but not the obligation to receive an exclusive license under the Licensed IP Rights to research, develop, make, have made, use, offer for sale, sell, have sold and import Products and otherwise exploit the Licensed IP Rights in the Territory in the Field, subject to the terms and conditions of this Agreement (the "Option"). During the Option Period, NEOMED shall not, without Artelo's prior written consent, directly or indirectly: (i) negotiate or enter into any agreement, arrangement or commitment according to which a Third Party is granted any right in the Territory under the Licensed IP Rights, (ii) take any action which may derogate from or conflict with, or refrain from taking any action which is necessary to preserve, the Option, or (iii) enter into any agreement, arrangement or commitment that would derogate from or conflict with the rights granted to Artelo under this Agreement.

3.2 Option Exercise. In order to exercise the Option, Artelo shall provide a written option exercise notice to NEOMED prior to the expiration of the Option Period (the "**Option Exercise Notice**"), and as of the date of the receipt of the Option Exercise Notice by NEOMED, the license grant to Artelo set forth in Section 4.1 shall automatically become effective. For the abundance of clarity, Artelo shall have the right to exercise the Option by providing the Option Exercise Notice to NEOMED at any point in time during the Option Period.

4. LICENSE GRANT

4.1 Licensed IP Rights. Conditional only upon Artelo serving the Option Exercise Notice and receipt of such Option Exercise Notice by NEOMED in accordance with Section 3.2, and subject to the terms and conditions of this Agreement, NEOMED hereby grants to Artelo an exclusive license under the Licensed IP Rights to research, develop, make, have made, use, offer for sale, sell, have sold and import Products and otherwise exploit the Licensed IP Rights in the Territory in the Field. During the term of this Agreement, NEOMED shall not enter into any agreement or otherwise license, grant, assign, transfer, convey or otherwise encumber or dispose any right, title or interest in, to or under any of the Licensed IP Rights, which agreement, license, grant, assignment, transfer, conveyance, encumbrance or disposition would conflict with the rights granted to Artelo hereunder.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 8 -

4.2 <u>Sublicenses</u>. Artelo may grant sublicenses under the license in Section 4.1 to Third Parties, provided that each such sublicense agreement is consistent with the terms and conditions of this Agreement. Artelo shall remain directly jointly and severally responsible for each of its sublicensees' compliance with this Agreement and shall promptly provide a copy of each such sublicense agreement to NEOMED, which sublicense agreement copy may be redacted, provided that no information necessary for NEOMED to verify the compliance of such sublicense agreement with this Agreement or to ascertain NEOMED's rights with respect thereto may be redacted.

4.3 Extension to Affiliates. Artelo shall have the right to extend the rights and obligations granted in this Agreement to one or more of its Affiliates. All applicable terms and provisions of this Agreement shall apply to any such Affiliate to which this Agreement has been extended to the same extent as such terms and provisions apply to Artelo. Artelo shall remain directly, jointly and severally liable for any acts or omissions of its Affiliates, and Artelo hereby expressly waives any requirement that NEOMED exhaust any right, power or remedy, or proceed directly against such Affiliate, for any obligation or performance hereunder prior to proceeding directly against Artelo.

4.4 <u>Rights Reserved</u>. This Agreement shall not be interpreted or construed as granting Artelo any rights, expressed or implied, to any of the Licensed IP Rights, or other form of rights other than the rights specifically and expressly granted herein and in accordance with this Agreement.

4.5 <u>Restriction</u>. In consideration for the rights granted herein by NEOMED, Artelo undertakes not to license, acquire or develop any rights relating to a compound of the same chemical class as the Compound (other than the Compound) during the Royalty Term.

4.6 IP Improvements. Subject to Sections 12.6.2 and 12.6.3, Artelo shall retain all rights, titles and interests in and to the IP Improvements.

5. CONSIDERATION

5.1 Option Consideration.

5.1.1 <u>Equity Grant</u>. On the Effective Date, as partial consideration for the grant of the Option by NEOMED to Artelo and the supply of the Technology Transfer Materials, Artelo shall grant NEOMED 120,000 fully paid and non-assessable shares of Artelo's common stock, subject to Artelo and NEOMED then executing a Common Stock Purchase Agreement in substantially the form attached hereto as Exhibit D (the "**Purchase Agreement**"). To the extent of any conflict between the terms of this Section 5.1.1 and the Purchase Agreement, the terms and conditions of the Purchase Agreement shall control and be determinative.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

-9-

5.1.2 <u>Cash Consideration</u>. As partial consideration for the grant of the Option by NEOMED to Artelo and the supply of the Technology Transfer Materials, Artelo shall make the following payments to NEOMED: (a) [***] due on the Effective Date, (b) [***], (c) [***], and (d) [***]; provided that if any of the foregoing payments have not accrued and become due prior to the earlier to occur of (i) the exercise of the Option by Artelo, and (ii) termination of this Agreement by Artelo pursuant to Section 12.2 hereof, then Artelo's payment obligations with respect to such payments that have not accrued and become null and void. If Artelo is unable to receive the Required Compound Supply within [***] after the Effective Date, then the due date for the payments owing under clauses (c) and (d) of this Section 5.1.2 will not be extended by more than [***].

5.2 License Consideration.

5.2.1 Equity Grant. Conditional upon Artelo's exercise of the Option, as partial consideration for the license granted to Artelo by NEOMED under Section 4.1, within ten (10) business days after the delivery of the Option Exercise Notice by Artelo to NEOMED, Artelo shall grant NEOMED a number of fully paid and non-assessable shares of Artelo's common stock equal to [***] of Artelo's Fully-Diluted Shares then outstanding, subject to Artelo and NEOMED then executing a Common Stock Purchase Agreement in substantially the form attached hereto as Exhibit E. For purposes of this Section 5.2.1, "**Fully-Diluted Shares**" shall mean, as of immediately prior to Artelo's exercise of the Option, the sum of (a) the outstanding shares of common stock of Artelo; (b) the shares of common stock of Artelo directly or indirectly issuable upon conversion or exchange of all outstanding options and warrants; and (c) the shares of common stock of Artelo reserved, but neither issued nor the subject of outstanding awards, under any equity incentive or similar plan of Artelo.

5.2.2 License Issuance Fee. Conditional upon Artelo's exercise of the Option, as partial consideration for the license granted to Artelo by NEOMED under Section 4.1, within ten (10) business days after the delivery of the Option Exercise Notice by Artelo to NEOMED, Artelo shall pay to NEOMED a license issuance fee of [***].

5.2.3 Royalties.

(a) <u>Royalty Rates</u>. Conditional upon Artelo's exercise of the Option, as partial consideration for the license granted to Artelo by NEOMED under Section 4.1, during the Royalty Term, Artelo shall pay royalties to NEOMED on Net Sales of Products by Artelo, its Affiliates and sublicensees in the Territory as follows:

Cumulative Net Sales ¹ in a Calendar Year (" <u>Annual Net Sales</u> ")	Rate
Portion of Annual Net Sales under or equal to \$500,000,000	[***]
Portion of Annual Net Sales greater than \$500,000,000	[***]

1. In calculating the Annual Net Sales for the purposes of determining the applicable royalty rate, Annual Net Sales shall include the cumulative Net Sales of Artelo, its Affiliates and sublicensees in a Calendar Year for all Products.



(b) <u>Royalty Payments</u>. Royalties will be payable on a calendar quarter basis. Within sixty (60) days after the end of each calendar quarter following the First Commercial Sale of the first Product, Artelo shall deliver to NEOMED a report containing the following information for the prior calendar quarter on a Product-by-Product and country-by-country basis: (i) the gross sales associated with each Product sold by Artelo and its Affiliates; (ii) a calculation of Net Sales of each Products that are sold by Artelo and its Affiliates; and (iii) a calculation of payments due to NEOMED with respect to the foregoing. Concurrently with these reports, Artelo shall remit to NEOMED any payment due for the applicable calendar quarter on Net Sales of Artelo and its Affiliates. Within seventy-five (75) days after the end of each calendar quarter following the First Commercial Sale of the first Product, Artelo shall deliver to NEOMED a report containing the following information for the prior calendar quarter on a Product-by-Product and country-by-country basis: (i) the gross sales associated with each Product sold by Artelo and its Affiliates. Within seventy-five (75) days after the end of each calendar quarter following the First Commercial Sale of the first Product, Artelo shall deliver to NEOMED a report containing the following information for the prior calendar quarter on a Product-by-Product and country-by-country basis: (i) the gross sales associated with each Product sold by Artelo's sublicensees; (ii) a calculation of Net Sales of each Product that are sold by Artelo's sublicensees; and (iii) a calculation of payments due to NEOMED with respect to the foregoing. Concurrently with these reports, Artelo shall remit to NEOMED any payment due for the applicable calendar quarter on Net Sales of Artelo's sublicensees. If no royalties are due to NEOMED for such reporting period, the report shall so state.

(c) <u>Royalty Adjustment</u>. In the event that the Royalty Rates payable to NEOMED negatively impacts the ability of Artelo, its Affiliates and sublicensees to maximize the Net Sales or earn reasonable profit thereon during the Royalty Term, but after the expiration of the last to expire Valid Claim in the applicable country, then the Parties agree to negotiate in good faith an adjustment to the royalty rates applicable to such country.

(d) <u>Combination Product</u>. If a Product is sold in a combination product with other pharmaceutically active components, Net Sales, for purposes of royalty payments on the combination product in a country, shall be calculated by multiplying the Net Sales of that combination product by the fraction A/A+B, where A is the invoice price of the Product sold separately in such country and B is the invoice price of the other active components in such combination product in such country. If no such separate sales are made by Artelo, its Affiliates or sublicensees, Net Sales for royalty determination shall be calculated by multiplying Net Sales of the combination by the fraction C/(C+D), where C is the fully allocated cost of the Product in such country and D is the fully allocated cost of such other active components in such country. Notwithstanding the above, Artelo shall in all cases notify and provide supporting data to NEOMED and NEOMED shall have the right to review any Combination Product calculations.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 11 -

(e) <u>Third Party Royalties</u>. If Artelo, its Affiliates or sublicensees are required to pay royalties to any Third Party in order to exercise its rights hereunder to develop, make, use, offer for sale, sell or import any Product, then Artelo shall have the right to credit up to [***] of such Third Party royalty actually paid against the royalties owing to NEOMED under Section 5.2.3(a) above with respect to sales of such Product; provided, however, that Artelo shall first inform NEOMED and not reduce the amount of the royalties paid to NEOMED under Section 5.2.3(a) above, with respect to sales of such Product, by more than [***].

5.2.4 <u>Milestone Payments</u>. Conditional upon Artelo's exercise of the Option, as partial consideration for the license granted to Artelo by NEOMED under Section 4.1, Artelo shall pay to NEOMED the following milestone payments:

[***]

Milestone payments owing by Artelo to NEOMED pursuant to Section 5.2.4(a) through (c) shall be payable by Artelo within [***] following the achievement of the corresponding milestone event. Milestone payments owing by Artelo to NEOMED pursuant to Section 5.2.4(d) through (g) shall be payable by Artelo within [***] following the achievement of the corresponding milestone event. For the avoidance of doubt, each milestone payment is only payable once, regardless of the number of times such milestone may be achieved by Artelo, its Affiliates and sublicensees, except that with respect to the milestone event set forth in Section 5.2.4(c), a corresponding milestone payment shall be due each time such milestone event is achieved. For the abundance of clarity, no milestone payment shall be due upon the Regulatory Approval of a Product for the first Indication in any country. For further clarity, all Annual Net Sales made by Artelo, its Affiliates and sublicensees prior to the expiration of the Royalty Term shall be considered in determining whether the milestone payments owing by Artelo to NEOMED pursuant to Section 5.2.4(d) through (g) become due.

Solely with respect to the milestone payment due under Section 5.2.4(a), payment shall be by any of the following, or a combination thereof, at the election of Artelo: (x) cash; or (y) issuance of shares of Artelo's common stock which shall be valued at its Fair Market Value on the date of exercise. For purposes of this Section 5.2.4, "Fair Market Value" shall mean (i) if Artelo's common stock is listed on any established stock exchange or a national market system, including without limitation the New York Stock Exchange, the NASDAQ Global Select Market, the NASDAQ Global Market or the NASDAQ Capital Market of The NASDAQ Stock Market, its Fair Market Value will be based on the five (5) day volume-weighted average price immediately preceding the milestone as quoted on such exchange or system on the day of determination, as reported in The Wall Street Journal or such other source as Artelo deems reliable; or (ii) in the absence of an established market for Artelo's common stock, the Fair Market Value will be determined in good faith by Artelo's board of directors provided that, the Fair Market Value will in no circumstances be greater than the price paid by a third party investor in the last private financing round in excess of \$1,000,000 completed by Artelo.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 12 -

5.2.5 <u>Strategic Transaction Consideration</u>. Conditional upon Artelo's exercise of the Option, as partial consideration for the license granted to Artelo by NEOMED under Section 4.1, Artelo shall pay to NEOMED a percentage of the Strategic Transaction Consideration in addition to the consideration due under Sections 5.2.1, 5.2.2, 5.2.3 and 5.2.4 as follows:

(a) [***] of the Strategic Transaction Consideration resulting from a Strategic Transaction consummated during the period from [***]; and
(b) [***] of the Strategic Transaction Consideration resulting from a Strategic Transaction consummated during the period from [***]; and

(c) [***] of the Strategic Transaction Consideration resulting from a Strategic Transaction consummated during the

period from [***].

Each Strategic Transaction Consideration payment owing by Artelo to NEOMED pursuant to this Section 5.2.5 shall be payable by Artelo within [***] following the receipt by Artelo of the corresponding Strategic Transaction Consideration. For clarity, no portion of any Strategic Transaction Consideration resulting from a Strategic Transaction consummated after [***] of the date of the Option Exercise Notice shall be due to NEOMED.

As used herein, "**Strategic Transaction**" means either (x) a grant by Artelo of a sublicense under the Licensed IP Rights to a Third Party (**'Sublicense**"), or (y) a transaction or a series of related transactions that results in an acquisition of Artelo by a Third Party, including by way of merger, purchase of capital stock or purchase of assets or change of control or otherwise ("**Acquisition**"); and "**Strategic Transaction Consideration**" means any cash consideration and the fair market value of any non-cash consideration paid to Artelo by (x) any sublicensee in consideration for a grant of a Sublicense under the Licensed IP Rights, including, without limitation upfront payments and milestone payments, but expressly excluding (i) royalties for sales of products or services, (ii) payments for the performance of or reimbursement for research or development activities performed by or on behalf of Artelo, (iii) payments as reimbursement for patent-related costs, (iv) payments for grants of rights to technology other than Licensed IP Rights, (v) payments for the supply of products, services or materials to a sublicensee, or (y) any acquirer as consideration for the assignment of this Agreement pursuant to an Acquisition, provided, however, that if contemporaneously with an assignment of this Agreement pursuant to an Acquisition, Artelo also assigns or licenses to the assignee other intellectual property or any other assets of Artelo, the fair market value of such other intellectual property and assets. No payment shall be determined pro rata based on the fair market value of the Licensed IP Rights relative to the fair market value of such other intellectual property and assets. No payment shall be due to NEOMED on any Strategic Transaction Consideration for a Sublicense (but excluding any Third Party that becomes an Affiliate of Artelo pursuant to the Acquisition).

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 13 -

6. ROYALTY REPORTS AND ACCOUNTING

6.1 <u>Royalty Reports</u>. During the term of this Agreement following the First Commercial Sale of a Product, Artelo shall furnish to NEOMED quarterly written reports showing in reasonably specific detail the calculation of royalties owing for the reporting period in accordance with Section 5.2.3(b). With respect to sales of Products invoiced in United States dollars, all amounts shall be expressed in United States dollars. With respect to sales of Products invoiced the United States dollars, all amounts shall be expressed in the domestic currency of the party making the sale together with the United States dollar equivalent. The United States dollar equivalent shall be calculated using the average of the exchange rate (local currency per US\$1) published in The Wall Street Journal, Western Edition, under the heading "Currency Trading" on the last business day of each month during the applicable calendar year. Artelo shall keep complete and accurate records in sufficient detail to enable the royalties and other payments payable hereunder to be determined.

6.2 Audits.

6.2.1 Upon the written request of NEOMED and not more than once in each calendar year, Artelo shall permit an independent certified public accounting firm of nationally recognized standing selected by NEOMED and reasonably acceptable to Artelo, at NEOMED's expense, to have access during normal business hours to such of the records of Artelo as may be reasonably necessary to verify the accuracy of the royalty reports for any year ending not more than three (3) years prior to the date of such request. The accounting firm shall disclose to NEOMED only whether or not the reports are correct and the amount of any discrepancies. No other information shall be shared.

6.2.2 If such accounting firm concludes that additional payments were owed during such period, Artelo shall make such additional payments including a yearly interest rate of [***] to be computed daily, within thirty (30) days of the date NEOMED delivers to Artelo such accounting firm's written report so concluding the fees and expenses charged by such accounting firm shall be paid by NEOMED; provided, if the audit correctly discloses that the amounts payable by Artelo for the audited period are [***] or more than the amounts actually paid by Artelo for such period, then Artelo shall pay the additional payments owed and the fees and expenses charged by such accounting firm.

6.2.3 NEOMED shall treat all financial information subject to review under this Section 6.2 as confidential, and shall cause its accounting firm to retain all such financial information in confidence under Section 10 below.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 14 -

7. PAYMENTS

7.1 <u>Payment Terms</u>. Royalties shown to have accrued by each royalty report provided for under Section 6.1 shall be due on the date such royalty report is due. The method of payment shall be by check or wire transfer to an address or account specified in writing by NEOMED. Payment in whole or in part may be made in advance of such due date.

7.2 Exchange Control. If at any time legal restrictions prevent the prompt remittance of part or all royalties with respect to any country in the Territory where the Product is sold, Artelo shall have the right, in its sole discretion, to make such payments by depositing the amount thereof in local currency to NEOMED's account in a bank or other depository institution in such country. If the royalty rate specified in this Agreement should exceed the permissible rate established in any country, the royalty rate for sales in such country shall be adjusted to the highest legally permissible or government-approved rate.

7.3 <u>Withholding Taxes</u>. Artelo shall be entitled to deduct the amount of any withholding taxes, value-added taxes or other taxes, levies or charges with respect to such amounts, other than United States taxes, payable by Artelo, its Affiliates or sublicensees, or any taxes required to be withheld by Artelo, its Affiliates or sublicensees, to the extent Artelo, its Affiliates or sublicensees pay to the appropriate governmental authority on behalf of NEOMED such taxes, levies or charges. Artelo shall use reasonable efforts to minimize any such taxes, levies or charges required to be withheld on behalf of NEOMED by Artelo, its Affiliates or sublicensees. Artelo promptly shall deliver to NEOMED proof of payment of all such taxes, levies and other charges, together with copies of all communications from or with such governmental authority with respect thereto.

8. DEVELOPMENT AND COMMERCIALIZATION OF PRODUCTS

8.1 <u>Option Period</u>. During the Option Period, prior to the exercise of the Option by Artelo, Artelo shall use Commercially Reasonable Efforts to conduct the activities set forth in the Research Plan at Artelo's sole expense. Artelo shall further provide NEOMED with updates on the progress of the Research Plan for each quarterly period during the Option Period in the form of an update phone call promptly following the conclusion of each of the first two (2) months of each quarterly period, and a summary written report provided promptly following the conclusion of each quarterly period.

8.2 <u>Following Option Exercise</u>. Following the exercise of the Option by Artelo, Artelo shall use Commercially Reasonable Efforts to research, develop and commercialize Products in the Territory in accordance with this Section 8.2. The efforts of Artelo's Affiliates and sublicensees shall be treated as the efforts of Artelo when evaluating Artelo's compliance with the foregoing diligence obligations. Without limiting the generality of the foregoing, Artelo will be responsible for conducting all necessary studies, including safety studies and clinical trials that are necessary in connection with seeking Regulatory Approvals to market the Product in the Territory, at Artelo's own cost and discretion.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 15 -

8.2.1 <u>Commercialization Efforts</u>. Artelo shall use Commercially Reasonable Efforts to obtain Regulatory Approval of at least one Product in each of the Major Market Countries. Promptly following the Regulatory Approval of a Product in a country in the Territory, Artelo shall use Commercially Reasonable Efforts to launch the Product in such country. Marketing, distribution and sale of the Products in the Territory shall be the sole responsibility of Artelo, which shall have the sole right and discretion to make all decisions relating thereto. Following launch of a Product in a country, Artelo shall use Commercially Reasonable Efforts to market and sell such Product in such country. The efforts of Artelo's Affiliates and sublicensees shall be treated as the efforts of Artelo when evaluating Artelo's compliance with the foregoing diligence obligations.

8.2.2 <u>Reporting</u>. Artelo shall provide NEOMED with written reports detailing the activities of Artelo, its Affiliates and sublicensees with respect to the research and development of and pre-commercial launch activities for Products in the Field in the Territory, both as to activities conducted during the prior period and planned activities for the Compound and Products, in sufficient depth to enable NEOMED to reasonably assess Artelo's compliance with its diligence obligations hereunder. Artelo shall present such report to NEOMED in conjunction with a meeting (either in person or by videoconference, as the Parties may agree) with Artelo's personnel responsible for the conduct of such development (and, if applicable, pre-commercial launch activities) which personnel shall include at least one Artelo representative responsible for such development (and, if applicable, pre-commercial launch activities) at a level of vice president or higher. Such reports shall be made on a semi-annual basis (within sixty (60) days following the end of each six (6) month period ending June 30 and December 31) until the First Commercial Sale occurs. In addition to such reports, upon reasonable request of NEOMED, Artelo will provide written summary updates of its progress since the close of the period covered by the last report provided.

8.2.3 Specific Diligence Obligations. Without limiting the generality of Sections 8.1 and 8.2, Artelo shall:

(a) Submit to NEOMED, within [***] following the delivery of the Option Exercise Notice, a certification supported by reasonable documentation, that Artelo, its Affiliates and sublicensees (or any combination thereof) control sufficient resources to maintain the performance of the obligations under this Section 8.2;

(b) Use Commercially Reasonable Efforts to initiate, directly or through its Affiliates or sublicensees, a Phase II Clinical Study for a Product within [***] following the delivery of the Option Exercise Notice; and

(c) Use Commercially Reasonable Efforts to initiate, directly or through its Affiliates or sublicensees, a Phase III Clinical Study for a Product within [***] following the delivery of the Option Exercise Notice.

8.2.4 <u>Regulatory Approvals and Regulatory Reporting</u>. Artelo will be responsible for the preparation and filing of the Regulatory Approvals for the Products with the applicable Regulatory Authorities in the Territory. Artelo shall prepare and file the Regulatory Approval Applications for the Products with the Regulatory Authorities in its name and at its cost. Artelo shall file, in its own name and at its own expense, all other applications for any approvals required for any clinical study or other study or action necessary or desirable to obtain such Regulatory Approval. Artelo shall have the sole responsibility for communicating with any Regulatory Authority regarding any Regulatory Approval Application or any Regulatory Approval for the Products once granted or any such other applications. Artelo shall be responsible for filing, at its own expense, all reports required to be filed in order to maintain any Regulatory Approvals granted for the Products.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 16 -

8.2.5 <u>Product Labeling</u>. Artelo shall be solely responsible for the preparing, updating and maintaining product labeling in connection with commercialization of the Products and in compliance with all applicable laws and regulations. Such labeling may include but is not limited to text and graphical contents of printed labels and labeling components, including but not necessarily limited to healthcare professional leaflets or inserts, patient leaflets or inserts, and cartons.

9. REPRESENTATIONS AND WARRANTIES

9.1 Mutual Representations and Warranties. Each Party hereby represents and warrants to the other Party as follows as of the Effective

Date:

9.1.1 Corporate Existence. Such Party is a not-for-profit corporation or a corporation, as applicable, duly organized, validly existing and in good standing under the laws of the jurisdiction in which it is organized.

9.1.2 <u>Authorization and Enforcement of Obligations</u>. Such Party (a) has the organizational power and authority and the legal right to enter into this Agreement and to perform its obligations hereunder, and (b) has taken all necessary organizational action on its part to authorize the execution and delivery of this Agreement and the performance of its obligations hereunder. This Agreement has been duly executed and delivered on behalf of such Party, and constitutes a legal, valid, binding obligation, enforceable against such Party in accordance with its terms.

9.1.3 <u>No Consents</u>. All necessary consents, approvals and authorizations of all governmental authorities and other Persons required to be obtained by such Party in connection with this Agreement have been obtained.

9.1.4 <u>No Conflict</u>. The execution and delivery of this Agreement and the performance of such Party's obligations hereunder (a) do not conflict with or violate any requirement of applicable laws or regulations, and (b) do not conflict with, or constitute a default under, any contractual obligation of it.

9.2 Additional NEOMED Representations and Warranties. In addition, NEOMED hereby represents and warrants to Artelo that:

9.2.1 Each item of the Licensed Patent Rights set forth in Exhibit A (a) is subsisting and in full force and effect, (b) has not been abandoned or passed into the public domain and (c) is free and clear of any liens or encumbrances (other than payment obligations to one or more Third Party).

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 17 -

9.2.2 NEOMED has not transferred ownership of, or granted any license of or right to use, or authorized the retention of any rights to use or joint ownership of, any Licensed IP Rights to any Person in any manner that would conflict with the license granted to Artelo in Section 4.1.

hereunder.

9.2.3 NEOMED exclusively Controls the Licensed IP Rights and has all rights necessary to grant the licenses to Artelo

9.2.4 No patent application or registration within the Licensed Patent Rights is the subject of any pending interference, opposition, cancellation or patent protest.

9.2.5 No Person has made any claim or allegation to NEOMED or its Affiliates in writing that such Person has any right or interest in, to or under the Licensed IP Rights.

9.2.6 Except as disclosed by NEOMED to Artelo prior to the Effective Date and summarized on Schedule 9.2.6 attached hereto, NEOMED has no knowledge of any facts, circumstances or information that (a) would render any Licensed Patent Right invalid or unenforceable or (b) would adversely affect any pending application for any Licensed Patent Right.

9.2.7 NEOMED has not misrepresented, or failed to disclose, and has no knowledge of any misrepresentation or failure to disclose, any material fact or circumstances in any application for any Licensed Patent Right that would constitute fraud or a misrepresentation with respect to such application or that would otherwise affect the validity or enforceability of any Licensed Patent Right.

9.2.8 All necessary registration, maintenance and renewal fees in connection with each item of the Licensed Patent Rights have been paid and to NEOMED's knowledge all necessary documents and certificates in connection with such Licensed Patent Rights have been filed with the relevant patent or other authorities in the United States or foreign jurisdictions, as the case may be, for the purposes of maintaining such Licensed Patent Rights.

9.2.9 No claim or litigation has been brought against it or to its knowledge threatened by any Third Party alleging that (a) the Licensed Patent Rights are invalid or unenforceable or (b) the manufacture, sale, offer for sale, importation or exploitation of the Compound and/or any subject matter covered by the Licensed IP Rights infringe or misappropriate or would infringe or misappropriate any right of any Third Party.

9.3 Disclaimers and Limitations

9.3.1 SUBJECT TO THE WARRANTIES SET FORTH IN SECTION 9.2, THE RIGHTS TO THE LICENSED IP RIGHTS GRANTED HEREUNDER ARE PROVIDED TO ARTELO ON AN "AS IS" BASIS IN ALL OTHER RESPECTS.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 18 -

9.3.2 EXCEPT AS EXPRESSLY PROVIDED FOR IN THIS ARTICLE 9, NEOMED DISCLAIMS ANY AND ALL REPRESENTATIONS AND WARRANTIES OF ANY KIND, EXPRESS OR IMPLIED, WITH RESPECT TO THE SUBJECT MATTER OF THIS AGREEMENT, INCLUDING, WITHOUT LIMITATION, WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR USE OR PURPOSE, NON-INFRINGEMENT AND ANY WARRANTY ARISING OUT OF PRIOR COURSE OF DEALING AND/OR USAGE OF TRADE.

9.3.3 WITHOUT LIMITING THE GENERALITY OF THE FOREGOING, NEOMED SPECIFICALLY DOES NOT MAKE ANY WARRANTIES OR REPRESENTATIONS EXPRESS OR IMPLIED, CONCERNING THE LICENSED IP RIGHTS OR OF THE COMPOUND OR PRODUCT OR THE SCOPE OF THE LICENSED PATENT RIGHTS. NEOMED MAKES NO REPRESENTATIONS, EXTENDS NO WARRANTIES, EXPRESS OR IMPLIED, AND ASSUMES NO LIABILITIES OR RESPONSIBILITIES WITH RESPECT TO THE USE, SALE OR OTHER DISPOSITION BY ARTELO, ITS AFFILIATES, ITS SUB-LICENSEES, ASSIGNEES OR END USERS OF THE COMPOUND OR THE PRODUCT MANUFACTURED OR DEVELOPED USING THE LICENSED IP RIGHTS. IN NO EVENT SHALL NEOMED BE LIABLE FOR ANY DIRECT, INCIDENTAL, SPECIAL OR CONSEQUENTIAL DAMAGES, INCLUDING LOSS OF PROFITS, REVENUE, DATA OR USE, INCURRED BY ARTELO, ITS AFFILIATES, OR THE SUBLICENSEES OR BY ANY OTHER PERSON, WHETHER IN AN ACTION IN CONTRACT OR TORT, ARISING UNDER THIS AGREEMENT OR IN RELATION WITH THE LICENSED IP RIGHTS, THE COMPOUND OR THE PRODUCT, EVEN IF NEOMED OR ANY THIRD PARTY HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES.

10. CONFIDENTIALITY

10.1 <u>Confidentiality Obligations</u>. Recipient shall keep and hold Confidential Information of Discloser in strictest confidence, and shall not use such Confidential Information for any purpose, other than as may be reasonably necessary for the performance of its duties under this Agreement, without Discloser's prior written consent. Recipient shall not disclose any such Confidential Information to any person or entity without Discloser's prior written consent. Recipient shall not disclose any such Confidential Information to any person or entity without Discloser's prior written consent, except to its and its Affiliates' employees, consultants and agents, as necessary for the sole purpose of performing Recipient's duties hereunder, under written terms and conditions no less protective of the Confidential Information than the terms and conditions of this Section 10. The obligations of confidentially under this Section 10 shall last until the applicable Confidential Information is no longer secret and confidential, through no fault of Recipient or any of its Affiliates or sublicensees, or until one of the exceptions in Section 1.5 applies to such Confidential Information, whichever occurs first.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 19 -

10.2 <u>Permitted Disclosures</u>. Notwithstanding anything herein to the contrary, Recipient may disclose Confidential Information of Discloser to the extent necessary to: (a) comply with an applicable law, regulation of a governmental agency or order of a court of competent jurisdiction, (b) to disclose information to any governmental agency for purposes of obtaining approval to test or market a Product or (c) prosecute or defend litigation; provided that if Recipient is required by law or regulation to make any such disclosure of Discloser's Confidential Information, it will give reasonable advance notice to Discloser of such disclosure requirement and will use commercially reasonable efforts to assist such Confidential information required to be disclosed. In addition, notwithstanding anything herein to the contrary, Recipient may disclose Discloser's Confidential Information to the extent (and only to the extent) such disclosure is reasonably necessary in the following instances: (i) in order for it to reasonably fulfill its obligations herein and to conduct its ordinary course of business, to its subcontractors, vendors, outside legal counsel, accountants and auditors under written obligations of confidential property rights in connection with Recipient's rights and obligations pursuant to this Agreement; and (iii) in connection with exercising its rights hereunder, to its Affiliates, potential and buffuture bona fide collaborators (including sublicensees, potential and permitted acquirers or assignees and potential investment bankers, investors and lenders) under written obligations of confidential information than the terms and conditions of confidentiality and non-use no less protective of the of the Confidential and permitted acquirers or assignees and potential investment bankers, investors and lenders) under written obligations of confidential investment bankers, investors and lenders) under written obligations of confidential information than the terms and conditions of confidentiality and non-use no le

10.3 <u>Confidential Terms</u>. Each of the Parties agrees not to disclose to any Third Party the terms and conditions of this Agreement without the prior approval of the other party, except to advisors (including financial advisors, attorneys and accountants), under written obligations of confidentiality and non-use no less protective of the Confidential Information than the terms and conditions of this Section 10, to potential and existing bona fide investors, financing sources, merger or other business partners and acquirers, on a need to know basis. Notwithstanding the foregoing, the Parties agree to issue, on such date as reasonably requested by Artelo, a joint press release to announce the execution of this Agreement, in the form mutually agreed to by the parties; thereafter, each party may each disclose to third parties the information contained in such press release without the need for further approval by the other party.

10.4 <u>SEC or Similar Filings</u>. Either Party may disclose the terms of this Agreement and events related to the development or commercialization of Products (including the receipt of milestone payments) to the extent reasonably required to comply with applicable laws, rules and regulations, including, without limitation, the rules and regulations promulgated by the United States Securities and Exchange Commission comparable foreign regulator and self-regulatory organizations (such as securities exchanges).

10.5 Injunctive Relief Authorized. Each Party as a Recipient acknowledges and agrees that (a) the Confidential Information of Discloser is of a special, unique, unusual, extraordinary and intellectual character; (b) the unauthorized use or disclosure of any Confidential Information of Discloser would constitute a material breach of this Agreement; (c) the interests of Discloser in and to the Confidential Information would be irreparably injured by the unauthorized use or disclosure of such information; and (d) money damages would not be sufficient to compensate Discloser for any such unauthorized use or disclosure. Accordingly, Recipient agrees that, in addition to any other remedies available to Discloser at law, in equity or under this Agreement, Discloser shall be entitled to seek specific performance, injunctive relief and other equitable relief to prevent any actual or threatened use or disclosure of the Confidential Information of Discloser without obligation to post any bond.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 20 -

11. PATENTS

11.1 Prosecution and Maintenance of Licensed Patent Rights

11.1.1 Option Period.

(a) Unless and until Artelo provides NEOMED with the Option Exercise Notice pursuant to Section 3.2, NEOMED shall be responsible for the preparation, filing, prosecution and maintenance of the Licensed Patent Rights in accordance with the provisions of this Section 11.1.1.

(b) NEOMED will diligently prosecute and maintain the United States and foreign patent applications and patents within the Licensed Patent Rights, subject to Artelo's reimbursement of all NEOMED's costs under Section 11.1.1(d). NEOMED will have sole responsibility for retaining and instructing patent counsel. NEOMED will provide Artelo with copies of all proposed patent office filings and correspondence, sufficiently in advance of submitting same to the patent office, so that Artelo may comment upon such documentation. NEOMED will promptly provide Artelo with copies of all official patent office correspondence so that Artelo may be informed of the continuing prosecution, and Artelo agrees to keep this documentation confidential in accordance with Section 10. Artelo may comment upon such documentation, provided that, if Artelo has not commented upon such documentation in reasonable time for NEOMED to sufficiently consider Artelo's comments prior to the deadline for filing a response with the relevant government patent office, NEOMED will be free to respond appropriately without consideration of Artelo's comments.

(c) NEOMED will cooperate with Artelo in good faith to prepare or amend any patent application within Licensed Patent Rights to include claims reasonably requested by Artelo to protect the Products contemplated to be developed by Artelo under this Agreement.

(d) Artelo shall reimburse NEOMED for all its costs for preparing, filing, prosecuting, and maintaining all United States and foreign patent applications and patents under Licensed Patent Rights that are incurred during the period from the Effective Date until either the patent prosecution and maintenance responsibilities are transitioned to Artelo pursuant to Section 11.1.2 or this Agreement is terminated. Patent cost reimbursement payments are due within thirty (30) days after receipt of invoice from NEOMED.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 21 -

11.1.2 Following Option Exercise. At such time as Artelo provides NEOMED with the Option Exercise Notice pursuant to Section 3.2, Artelo shall assume and shall be responsible for the preparation, filing, prosecution and maintenance of the Licensed Patent Rights, at Artelo's sole expense. Promptly following NEOMED's receipt of the Option Exercise Notice, and in any event within thirty (30) days thereafter, the Parties shall cooperate in good faith to execute a transition of the patent prosecution and maintenance responsibilities from NEOMED to Artelo. NEOMED shall cause its patent prosecution counsel to cooperate with Artelo's patent prosecution counsel to ensure a smooth transition. Artelo will provide NEOMED may comment upon such documentation. Artelo will promptly provide NEOMED with copies of all proposed patent office filings and correspondence, sufficiently in advance of submitting same to the patent office, so that Artelo may comment upon such documentation. Artelo will promptly provide NEOMED with copies of all official patent office correspondence so that Artelo may be informed of the continuing prosecution. If Artelo elects not to file any patent application claiming any Licensed Patent Rights or otherwise abandon the prosecution and maintenance of any patent application or patent Rights, then (a) Artelo shall be given no later than thirty (30) days prior to the next deadline to take any necessary action with the relevant patent office); and (b) NEOMED shall have the right but not the obligation to control the filing, prosecution and maintenance of such actions or patent application or patent application or patent rights, at NEOMED's sole cost.

11.2 Enforcement of Licensed Patent Rights.

11.2.1 Each Party shall notify the other Party of any infringement known to such Party of any Licensed Patent Rights by a Third Party and shall provide the other Party with the available evidence, if any, of such infringement.

11.2.2 Artelo, at its sole expense, shall have the first right to determine the appropriate course of action to enforce the Licensed Patent Rights or otherwise abate such infringement, to take (or refrain from taking) appropriate action to enforce the Licensed Patent Rights, to control any litigation or other enforcement action and, subject to NEOMED's written consent, not to be unreasonably withheld, delayed or conditioned, to enter into, or permit, the settlement of any such litigation or other enforcement action with respect to the Licensed Patent Rights, and shall endeavor, in good faith, that the interests of NEOMED are protected in so doing. NEOMED agrees to cooperate reasonably with Artelo in any action to enforce the Licensed Patent Rights under this Section 11.2.2, provided that Artelo reimburses NEOMED promptly for any costs and expenses incurred by NEOMED in providing such assistance. Without limiting the generality of the foregoing, should Artelo elect to bring suit against an infringer of Licensed Patent Rights, NEOMED agrees to be joined as party plaintiff in any such suit, if deemed a necessary party, subject to Artelo's reimbursement of NEOMED's costs and expenses as set forth above.

11.2.3 If Artelo does not, within [***] of receipt of notice from NEOMED under Section 11.2.1, abate the infringement or file suit to enforce the Licensed Patent Rights, NEOMED shall have the right to take whatever action it deems appropriate to enforce the Licensed Patent Rights. The Party controlling any such enforcement action shall not settle the action or otherwise consent to an adverse judgment in such action that adversely affects the rights or interests of the non-controlling Party without the prior written consent of the other Party.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 22 -

11.2.4 All monies recovered upon the final judgment or settlement of any such suit to enforce the Licensed Patent Rights pursuant to this Section 11.2 shall be first used to reimburse each Party for its costs and expenses incurred in connection with such suit pro rata, and the remainder, if any, shall be allocated as follows: (a) for any suit initiated and prosecuted by Artelo, the remainder shall be deemed Net Sales and allocated in accordance with Section 5.2.3, and (b) for any suit initiated and prosecuted by NEOMED, the remainder shall be remitted to NEOMED.

11.3 IP Improvements. Artelo shall be responsible for the preparation, filing, prosecution, maintenance and enforcement of the patent rights within the IP Improvements, at Artelo's sole expense

11.4 Undertakings.

11.4.1 Artelo hereby undertakes not to challenge the validity of any of the Licensed IP Rights, nor institute any legal proceeding of any nature and kind against NEOMED, its Affiliates or sublicensees in respect of the Compounds, the Licensed IP Rights and/or Products relating to this Agreement, other than with respect to any uncured breach of this Agreement by NEOMED.

11.4.2 Artelo hereby agrees to comply with the patent marking laws and regulations in each country of the Territory in which any Product is sold by Artelo, its Affiliates or its sublicensees.

12. TERMINATION

12.1 Expiration. Subject to early termination pursuant to the provisions of Sections 5.2.3 above. If this Agreement expires pursuant to clause (b) of this Section 12.1, then upon expiration of the Royalty Term, on a country-by-country basis, the licenses granted to Artelo by NEOMED under this Agreement shall be fully paid-up and irrevocable in the countries in which the Royalty Term has expired.

12.2 <u>Termination by Artelo</u>. Artelo may terminate this Agreement in its entirety, in its sole discretion, at any time effective (a) [***], during the Option Period, or (b) [***] after NEOMED's receipt of a written notice of termination provided by Artelo to NEOMED, following the date of the Option Exercise Notice.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 23 -

12.3 <u>Termination by NEOMED</u>. NEOMED shall have the right to terminate this Agreement if Artelo does not exercise Commercially Reasonable Efforts to meet its obligations under Section 8.2.3(a), (b) or (c) in accordance with the following procedure:

(a) If Artelo fails to meet its obligations under Section 8.2.3(a), (b) or (c), the Parties shall promptly review the underlying reasons and agree on appropriate measures to enable Artelo to meet such obligations. Artelo shall use Commercially Reasonable Efforts to pursue implementation of such measures in good faith during a cure period of [***]. Should Artelo (i) fail to use Commercially Reasonable Efforts to pursue implementation of such agreed appropriate measures in good faith and (ii) as a result fails to meet its obligations under Section 8.2.3(b) or (c) prior to the expiration of the [***] cure period, NEOMED shall have the right to terminate this Agreement upon [***] written notice.

(b) Notwithstanding the above, if any such failure by Artelo was due to a material adverse event or any factors which could have not been reasonably foreseen by Artelo or which were outside the control of Artelo (including without limitation any action or failure to act by any Regulatory Authority or any other governmental authority, including without limitation the FDA or the U.S. Drug Enforcement Administration), then (i) Artelo shall notify NEOMED of such event or factors promptly upon becoming aware thereof and (ii) a new planning process will be triggered to determine revised targets taking into account the estimated impact of such events and/or factors promptly following such notice.

12.4 <u>Termination for Cause</u>. Either Party will have the right to terminate this Agreement in full upon delivery of written notice to the other Party in the event of any material breach by the other Party of any terms and conditions of this Agreement, provided, that such termination will not be effective if such breach has been cured within [***] after written notice thereof is given by the non-breaching Party to the breaching Party specifying in reasonable detail the nature of the alleged breach; provided further, however, that if the material breach is not reasonably capable of being cured within the [***] cure period, and if the breaching party (a) proposes within such [***] a written plan to cure such breach that is reasonably acceptable to the non-breaching Party, and (b) makes good faith efforts to cure such default and to implement such written cure plan and reports at least monthly to the non-breaching Party is diligently pursuing such cure in accordance with such plan. Further, if the material breach is disputed, then no termination pursuant to this Section 12.4 shall be effective prior to the resolution of such dispute resolution process.

12.5 <u>Termination for Business Failure</u>. Either Party will have the right to terminate this <u>Agreement immediately upon written notice to</u> the other Party in the event that (i) the other Party becomes insolvent, or (ii) a receiver, trustee, or custodian is appointed for the other Party, or (iii) the other Party makes an assignment for the benefit of creditors, or is involuntary liquidated or dissolved, and (iv) in the event of the occurrence of any action or event which is, comparable in law of one or more of the events described in this Section 12.5.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 24 -

12.6 <u>Effect of Expiration or Termination</u>. Upon expiration of this Agreement pursuant to Section 12.1(a) (but not expiration of this Agreement pursuant to Section 12.1(b)) or any termination of this Agreement pursuant to Section 12.2 or 12.3 or any termination of this Agreement by NEOMED pursuant to 12.4 or 12.5:

12.6.1 <u>Termination of Licenses</u>. All rights and licenses granted to Artelo hereunder shall terminate.

12.6.2 <u>Transfer of Data and Compound-Related Improvements</u>. Artelo further agrees to transfer all Data and Compound-Related Improvements in its possession and control to NEOMED as soon as reasonably practicable.

12.6.3 <u>Product-Related Improvements</u>. To the extent requested in writing by NEOMED within [***] after the effective date of termination, the Parties shall negotiate, in good faith, an agreement on commercially reasonable terms, pursuant to which Artelo would grant to NEOMED a license under Artelo's intellectual property rights in the Product-Related Improvements to develop, make, use, sell, offer for sale and import the Compound or a Product. Such negotiation shall take into consideration amounts invested by Artelo in the clinical development of the Compound and Products.

12.6.4 <u>Confidential Information</u>. Recipient shall, within thirty (30) days after the effective date of termination and at Recipient's expense, return or destroy, at Discloser's election, all Discloser Know-How and other Confidential Information of Discloser (provided that Recipient may keep one copy of such Confidential Information for archival purposes only and such additional copies of or any computer records or files containing such Confidential Information that have been created solely by Recipient's automatic or routine archiving and back-up procedures, to the extent created and retained in a manner consistent with Recipient's standard archiving and back-up procedures, but not for any other use or purpose, subject to an ongoing obligation of confidentiality in accordance with Section 10).

12.6.5 <u>Sublicenses</u>. If this Agreement is terminated pursuant to Section 12.2, 12.3, 12.4 or 12.5, all sublicenses that are granted by Artelo pursuant to this Agreement and in accordance with Section 4.2 where the sublicensee is in compliance with its sublicense agreement and this Agreement as of the date of such termination will remain in effect and will be assigned to NEOMED, provided that [***].

12.7 <u>Survival</u>. Expiration or termination of this Agreement shall not relieve the parties of any obligation accruing prior to such expiration or termination, and the provisions of Sections 2.2.3, 2.2.4, 4.4, 4.6, 5.2.3, 6, 9.3, 10, 12.6, 12.7, 13 and 14, and such other Sections of this Agreement that are required to give effect to such Sections, shall survive the expiration or termination of this Agreement.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 25 -

13. INDEMNIFICATION

13.1 Indemnification by Artelo. Artelo shall defend, indemnify and hold NEOMED and its (and its Affiliates') members, directors, officers, employees and agents harmless from all losses, liabilities, damages and expenses (including reasonable attorneys' fees and costs) (collectively, "Losses") resulting from any claims, demands, actions and other proceedings including without limitation, for bodily injury (including death), to the extent resulting from Artelo's or its Affiliates' or sublicensees' use of the Licensed IP Rights, any use or handling of the Compound substance or research, development or commercialization of the Compound or any Product under this Agreement except to the extent such Losses are subject to NEOMED's indemnification obligations under Section 13.2 below.

13.2 Indemnification by NEOMED. NEOMED shall defend, indemnify and hold Artelo and its (and its' Affiliates') directors, officers, employees and agents harmless from all Losses resulting from any claims, demands, actions and other proceedings to the extent resulting from (a) the material breach of the representations and warranties made hereunder by NEOMED or its Affiliates or (b) any governmental charges, fines, penalties or similar amounts arising from any use or handling of the Compound substance by or on behalf of NEOMED or its Affiliates, except to the extent such Losses are subject to Artelo's indemnification obligations under Section 13.1 above.

13.3 Procedure. The Party seeking indemnification (the 'Indemnified Party") promptly shall notify the other party (the ''Indemnifying Party") of any claim, demand, action or other proceeding for which the Indemnified Party intends to claim indemnification. The Indemnifying Party shall have the right to participate in, and to the extent the Indemnifying Party so desires jointly with any other indemnitor similarly noticed, to assume the defense thereof with counsel selected by the Indemnifying Party; provided, however, that the Indemnified Party shall have the right to retain its own counsel, with the fees and expenses to be paid by the Indemnified Party. The indemnity obligations under this Section 13 shall not apply to amounts paid in settlement of any claim, demand, action or other proceeding if such settlement is effected without the prior express written consent of the Indemnifying Party, which consent shall not be unreasonably withheld or delayed. The failure to deliver notice to the Indemnified Party under this Section 13 with respect thereto, but the omission so to deliver notice to the Indemnifying Party shall not relieve it of any liability that it may have to the Indemnified Party other than under this Section 13 . The Indemnifying Party may not settle or otherwise consent to an adverse judgment in any such claim, demand, action or other proceeding, the Indemnifying Party shall not be unreasonably withheld or delayed. The Indemnified Party without the prior express written consent of the Indemnifying Party and its legal representatives in the indemnified Party, the indemnified Party, which consent shall not be unreasonably withheld or delayed. The Indemnified Party without the prior express written consent of the Indemnifying Party and its legal representatives in the investigation of any claim, demand, action or other proceeding covered by this Section 13 .

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 26 -

13.4 <u>LIMITATION OF DAMAGES</u>. IN NO EVENT SHALL EITHER PARTY BE LIABLE HEREUNDER TO THE OTHER PARTY FOR ANY PUNITIVE, INDIRECT, SPECIAL, INCIDENTAL OR CONSEQUENTIAL DAMAGES (INCLUDING LOST REVENUE, LOST PROFITS OR LOST SAVINGS) HOWEVER CAUSED AND UNDER ANY THEORY, EVEN IF IT HAS NOTICE OF THE POSSIBILITY OF SUCH DAMAGES. THE LIMITATIONS SET FORTH IN THIS SECTION 13.4 SHALL NOT APPLY WITH RESPECT TO (I) ANY BREACH OF SECTION 10 OR (II) THE INTENTIONAL MISCONDUCT OR GROSS NEGLIGENCE OF A PARTY. NOTHING IN THIS SECTION 13.4 IS INTENDED TO LIMIT OR RESTRICT THE INDEMNIFICATION RIGHTS OR OBLIGATIONS OF A PARTY UNDER THIS SECTION 13 WITH RESPECT TO ANY DAMAGES OWED OR PAID TO A THIRD PARTY IN CONNECTION WITH A THIRD PARTY CLAIM.

13.5 Insurance.

13.5.1 Artelo shall maintain at its own cost and at all times during the term of this Agreement policies of insurance consistent with normal business practices of prudent pharmaceutical companies similarly situated. Such insurance policies shall include, without limitation, commercial general liability insurance, including, without limitation, product liability, covering claims for damages because of bodily injury (including, without limitation, death), personal injury and property damage arising out of Artelo's acts or omissions and including coverage for contractual liabilities. Without limiting the foregoing, no later than Artelo's commencement of clinical trials in respect of any Product, Artelo shall obtain, and maintain in full force and effect, adequate clinical trials insurance, for claims arising out of Artelo, Artelo shall obtain, and maintain in full force and effect, adequate clinical trials insurance for obehalf of Artelo, Artelo shall obtain, and maintain in full force and effect, adequate for commercial distribution of any Product by or on behalf of Artelo, Artelo shall obtain, and maintain in full force and effect, adequate for bodily injury and property damage claims.

13.5.2 The policies described in Section 13.5.1 above will be in such amounts and cover such risks as are reasonable and prudent for those types of policies, but shall in no event be less than, in the aggregate: (a) [***] as of the Effective Date, (b) [***] as of the date of the commencement of any clinical trial, and (c) [***] as of the date of the First Commercial Sale. Such policies will be written by insurance companies with an A.M. Best's rating of A:VIII or higher (or if such policies are not subject to the Best rating, then by carriers who are reasonably acceptable to NEOMED). The foregoing policies will: (i) cover claims arising out of the performance of this Agreement that are made within a period of not less than six (6) years after the expiration or earlier termination of this Agreement; and (ii) be primary and non-contributory to any liability insurance carried by NEOMED, which insurance will be excess for claims and losses arising out of the performance of this Agreement. The policies described above will be specifically endorsed to list NEOMED as an additional insured (as long as such endorsement is allowed by Applicable Law), and Artelo will notify NEOMED at least thirty (30) days in advance of any cancellation or non-renewal of or material changes in of such insurance coverage. Artelo shall provide NEOMED with a valid, current certificate of insurance as evidence of the insurance required herein upon request. Maintenance of such insurance coverage will not relieve Artelo of any responsibility under this Agreement for damages in excess of insurance limits or otherwise.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 27 -

14. MISCELLANEOUS

14.1 <u>Notices</u>. Any consent, notice or report required or permitted to be given or made under this Agreement by one of the parties to the other shall be in writing and addressed to such other party at its address indicated below, or to such other address as the addressee shall have last furnished in writing to the addressor for purposes of this Section 14.1, and shall be effective upon receipt by the addressee.

If to NEOMED:	The NEOMED Institute
	7171 Frederick-Banting
	Saint-Laurent, Quebec H4S 1Z9, Canada
	Attention: Daniel Böck, VP Business Development
If to Artelo:	Artelo Biosciences, Inc.
	888 Prospect Street, Suite 210
	La Jolla, CA 92037 USA
	Attention : Gregory D. Gorgas, President & CEO

14.2 <u>Assignment</u>. Except as otherwise expressly provided under this Agreement neither this Agreement nor any right or obligation hereunder may be assigned or otherwise transferred (whether voluntarily, by operation of law or otherwise), without the prior express written consent of the other party; provided, however, that either party may, without such consent, assign this Agreement and its rights and obligations hereunder (i) to an Affiliate or (ii) in connection with the transfer or sale of all or substantially all of its business relating to this Agreement, or in the event of its merger, consolidation, change in control or similar transaction. In addition, NEOMED shall have the right to hypothecate, pledge, assign in warranty or otherwise charge its payment rights hereunder in favour of a financial institution as security. This Agreement shall be binding upon and inure to the benefit of each party and its successors and permitted assigns. Any purported assignment or transfer in violation of this Section REF _Ref486176120 \r \h *

14.3 <u>Governing Law; Jurisdiction; Venue</u>. This Agreement shall be governed by and construed in accordance with the laws of the State of New York, without regard to the conflicts of law principles thereof. For any legal action arising from or related to this Agreement, the Parties hereby irrevocably: (a) consent solely to personal jurisdiction of and exclusive venue in the state and federal courts located in New York County, New York, USA; (b) agree that such courts will be the sole courts utilized; and (c) hereby waive any jurisdictional or venue objections to such courts, including without limitation, forum non conveniens. If any dispute arises between the Parties in connection with this Agreement which leads to a proceeding to resolve such dispute, the prevailing Party in such proceeding will be entitled to receive its reasonable attorneys' fees, expert witness fees and out-of-pocket costs incurred in connection with such proceeding, in addition to any other relief it may be awarded.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 28 -

14.4 Entire Agreement. This Agreement contains the entire understanding of the Parties with respect to the subject matter hereof. There are no agreements, representations, warranties, covenants or undertakings with respect to the subject matter hereof other than those expressly set forth herein. All express or implied representations, agreements and understandings relating to such subject matter, either oral or written, heretofore made are expressly superseded by this Agreement, including, without limitation the Confidentiality Agreement.

14.5 <u>Independent Contractors</u>. Each Party hereby acknowledges that the Parties shall be independent contractors and that the relationship between the Parties shall not constitute a partnership, joint venture or agency. Neither Party shall have the authority to make any statements, representations or commitments of any kind, or to take any action, which shall be binding on the other Party, without the prior consent of the other Party to do so.

14.6 <u>Remedies</u>. All remedies, either under this Agreement, by law, or otherwise afforded to any Party, shall be cumulative and not alternative.

14.7 Force Majeure. Neither Party shall be held liable or responsible to the other Party nor be deemed to have defaulted under or breached this Agreement for failure or delay in fulfilling or performing any term of this Agreement solely to the extent, and for so long as, such failure or delay is caused by or results from causes beyond the reasonable control of the affected Party including but not limited to fire, floods, embargoes, war, acts of war (whether war be declared or not), insurrections, riots, civil commotions, strikes, lockouts or other labor disturbances, acts of God or acts, omissions or delays in acting by any governmental authority or the other Party. If any such event continues for more than ninety (90) days, then such other Party shall have the right to terminate this Agreement upon thirty (30) days prior written notice to the affected Party.

14.8 Fees and Expenses. Each Party shall pay its own costs and expenses in connection with this Agreement and the transactions contemplated hereby (including the fees and expenses of its advisers, accountants and legal counsel).

14.9 Section 365(n) of the Bankruptcy Code. All rights and licenses granted under or pursuant to any section of this Agreement are, and shall otherwise be deemed to be, for purposes of Section 365(n) of the U.S. Bankruptcy Code, licenses of rights to "intellectual property" as defined under Section 101(35A) of the U.S. Bankruptcy Code to the extent permitted thereunder. Artelo shall retain and may fully exercise all of its rights and elections under the U.S. Bankruptcy Code. Upon the bankruptcy of NEOMED, Artelo shall further be entitled to a complete duplicate of (or complete access to, as appropriate) any such intellectual property (solely to the extent that such license in effect at such time), and such, if not already in its possession, shall be promptly delivered to Artelo, unless NEOMED elects to continue, and continues, to perform all of its obligations under this Agreement.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 29 -

14.10 <u>Further Assurances</u>. At any time or from time to time after the date hereof, the Parties agree to cooperate with each other, and at the request of any other party, to execute and deliver any further instruments or documents and to take all such further action as the other Party may reasonably request in order to evidence or effectuate the consummation of the transactions contemplated hereby and to otherwise carry out the intent of the parties hereunder.

14.11 Interpretation. The captions to the Sections of this Agreement are not a part of this Agreement, but are included for convenience of reference and shall not affect its meaning or interpretation. In this Agreement: (i) the word "including," "includes," "included," and "include" shall be deemed to be followed by the phrase "without limitation" or like expression; (ii) the singular shall include the plural and *vice versa*; (iii) masculine, feminine, and neuter pronouns and expressions shall be interchangeable; (iv) the words "hereof," "herein," "hereto," "hereby," "hereunder," and derivative or similar words refer to this Agreement as an entirety and not solely to any particular provision of this Agreement; (v) each reference in this Agreement to a particular Section, appendix, schedule, or exhibit means a Section, appendix, schedule, or exhibit means a Section, appendix, schedule, or or this Agreement, unless another agreement is specified; (vi) "the word "will" shall be construed to have the same meaning and effect as the word "shall"; (vii) "or" is not disjunctive (i.e., it means "and/or") unless the context clearly requires otherwise; (viii) references to any part or Person shall include its permitted successors or assigns; and (ix) whenever this Agreement refers to a number of days, such number shall refer to calendar days unless business days are specified; and business days are specified; and business. This Agreement has been prepared jointly and shall not be strictly construed against either Party.

14.12 Waivers. It is agreed that no delay or omission to exercise any right, power or remedy accruing to any Party, upon any breach, default or noncompliance by the other Party under this Agreement, shall impair any such right, power or remedy, nor shall it be construed to be a waiver of any such breach, default or noncompliance, or any acquiescence therein, or of or in any similar breach, default or noncompliance thereafter occurring. It is further agreed that any waiver, permit, consent or approval of any kind or character on the part of any party hereto of any breach, default or noncompliance under this Agreement or any waiver on such party's part of any provisions or conditions of this Agreement, must be in writing and shall be effective only to the extent specifically set forth in such writing. The waiver by a Party of any right hereunder, or of any failure to perform or breach by the other Party hereunder whether of a similar nature or otherwise.

14.13 <u>Severability</u>. Whenever possible, each provision of this Agreement shall be interpreted in such manner as to be effective and valid under applicable law, but if any provision of this Agreement is held to be invalid, illegal or unenforceable in any respect under any applicable law or rule in any jurisdiction, such invalidity, illegality or unenforceability shall not affect any other provision or any other jurisdiction, but this Agreement shall be reformed, construed and enforced in such jurisdiction as if such invalid, illegal or unenforceable provision had never been contained herein.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- 30 -

14.14 Enforcement. Each Party hereto acknowledges that money damages would not be an adequate remedy in the event that any of the covenants or agreements in this Agreement are not performed by the Parties in accordance with its terms, and it is therefore agreed that in addition to and without limiting any other remedy or right each party may have, each party will have the right to an injunction, temporary restraining order or other equitable relief in any court of competent jurisdiction enjoining any such breach and enforcing specifically the terms and provisions hereof.

14.15 <u>Counterparts</u>. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. Counterparts may be delivered via facsimile, electronic mail (including pdf) or other transmission method and any counterpart so delivered shall be deemed to have been duly and validly delivered and be valid and effective for all purposes.

[Signature Page Follows.]

IN WITNESS WHEREOF, the parties have executed this Material and Data Transfer, Option and License Agreement as of the Effective Date.

NEOMED Institute

By: <u>(signed) Donald Olds</u> Donald Olds, President & CEO

Artelo Biosciences, Inc.

By: <u>(signed) Gregory D. Gorgas</u> Gregory D. Gorgas, President & CEO

[Signature Page to Material and Data Transfer, Option and License Agreement]

EXHIBIT A

LICENSED PATENT RIGHTS

[***]

EXHIBIT B

RESEARCH PLAN

EXHIBIT C

TECHNOLOGY TRANSFER MATERIALS

[***]

EXHIBIT D

PURCHASE AGREEMENT

ARTELO BIOSCIENCES, INC. COMMON STOCK PURCHASE AGREEMENT

This Common Stock Purchase Agreement (this "Agreement") is made as of January 2, 2018, by and between Artelo Biosciences, Inc., a Nevada corporation (the "Company"), and NEOMED Institute, a not-for-profit corporation established under the *Not-for-Profit Corporations Act* (Canada) (the "Purchaser").

Reference is hereby made to the Material and Data Transfer, Option and License Agreement dated as of the date hereof by and between the Company and the Purchaser (the "**Development Agreement**"). Capitalized terms used in this Agreement that are not otherwise defined herein shall have the respective meanings assigned to them in the Development Agreement.

In consideration of the mutual covenants and representations set forth below, the Company and Purchaser agree as follows:

1. Issuance of the Shares. Subject to the terms and conditions of this Agreement, as partial consideration for the grant of the Option (as defined in the Development Agreement) by NEOMED to Artelo and the supply of the Technology Transfer Materials (as defined in the Development Agreement) and pursuant to the terms of Section 5.1.1 of the Development Agreement, the Company hereby issues to the Purchaser on the Closing (as defined below) One Hundred and Twenty Thousand (120,000) shares of the Company's common stock (the "Shares").

2. Closing. The purchase and sale of the Shares shall occur at a closing (the 'Closing') to be held on the date hereof.

3. Restrictions on Transfer.

A. Investment Representations and Legend Requirements. The Purchaser hereby makes the investment representations listed on Exhibit A to the Company as of the date of this Agreement, and agrees that such representations are incorporated into this Agreement by this reference, such that the Company may rely on them in issuing the Shares. Purchaser understands and agrees that the Company shall cause the legends set forth below, or substantially equivalent legends, to be placed upon any certificate(s) evidencing ownership of the Shares, together with any other legends that may be required by the Company or by applicable state or federal securities laws:

THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 (THE "*ACT*") AND MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED, PLEDGED OR HYPOTHECATED UNLESS AND UNTIL REGISTERED UNDER THE ACT OR, IN THE OPINION OF COUNSEL SATISFACTORY TO THE ISSUER OF THESE SECURITIES, SUCH OFFER, SALE OR TRANSFER, PLEDGE OR HYPOTHECATION OTHERWISE COMPLIES WITH THE ACT.

THE SHARES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO CERTAIN RESTRICTIONS ON TRANSFER IN THE EVENT OF A PUBLIC OFFERING AS SET FORTH IN THE COMMON STOCK PURCHASE AGREEMENT BETWEEN THE ISSUER AND THE ORIGINAL HOLDER OF THESE SHARES, A COPY OF WHICH MAY BE OBTAINED AT THE PRINCIPAL OFFICE OF THE ISSUER. SUCH TRANSFER RESTRICTIONS ARE BINDING ON TRANSFEREES OF THESE SHARES.

B. *Stop-Transfer Notices*. Purchaser agrees that to ensure compliance with the restrictions referred to herein, the Company may issue appropriate "stop transfer" instructions to its transfer agent, if any, and that, if the Company transfers its own securities, it may make appropriate notations to the same effect in its own records.

C. *Refusal to Transfer*. The Company shall not be required (i) to transfer on its books any Shares that have been sold or otherwise transferred in violation of any of the provisions of this Agreement or (ii) to treat as owner of such Shares or to accord the right to vote or pay dividends to any purchaser or other transferee to whom such Shares shall have been so transferred.

D. *Limitation on restriction:* Notwithstanding anything contained in this Agreement (including Exhibit A), the Company acknowledges and agrees that the Purchaser has the right to transfer [***] at any time, provided that the terms and conditions of this Agreement (including Exhibit A) equally apply to [***].

4. *Tax Consequences.* The Purchaser has reviewed with the Purchaser's own tax advisors the federal, state, local and foreign tax consequences of this investment and the transactions contemplated by this Agreement. The Purchaser is relying solely on such advisors and not on any statements or representations of the Company or any of its agents. The Purchaser understands that the Purchaser (and not the Company) shall be responsible for any tax liability that may arise as a result of the transactions contemplated by this Agreement.

5. Representations and Warranties of the Company

The Company hereby makes the following representations and warranties to the Purchaser as of the date of this Agreement and acknowledges that the Purchaser is relying on such representations and warranties in entering into this Agreement:

A. the Company is a corporation duly formed and validly existing under the laws of Nevada and has all necessary corporate power and authority to execute this Agreement and any certificate evidencing ownership of the Shares (the "**Certificate**") and to offer, sell and issue the Shares, and to perform its obligations under this Agreement and the Certificate;

B. the authorized capital of the Company consists of 150,000,000 shares of common stock, of which 11,352,302 are currently issued and outstanding;

C. the Company has complied and will comply with all applicable corporate and securities laws and regulations in connection with the offer, sale and issuance of the Shares, and the execution, delivery and performance of this Agreement and the transactions contemplated by this Agreement will not result in any violation of or constitute either a default under any provision of its organizational documents or of any material agreement or instrument to which the Company is a party or by which the Company is bound, or any judgment, decree, order, law, statute, rule or regulation applicable to the Company;

D. this Agreement has been duly and validly authorized, executed and delivered by the Company and constitutes a legal, valid and binding obligation of the Company, enforceable against the Company in accordance with its terms;

E. the issuance of the Shares has been duly authorized and approved by all requisite corporate, regulatory and other action and upon Closing the Shares will be validly issued as fully paid and non-assessable, and shall be free and clear of any liens, mortgages, charges, security interests, burdens, encumbrances and other restrictions or limitations of any kind whatsoever;

F. the Company is duly qualified to transact business in each jurisdiction where the nature and extent of its business and properties (both owned and leased) requires;

G. the Company has been conducting its business in the ordinary course in compliance with all applicable laws, rules and regulations, including but not limited to federal, state, municipal and local laws, rules and regulations of each jurisdiction in which it carries on business and is not in breach of any such laws, rules or regulations;

H. the Company does not have any information or knowledge of any fact not communicated to the Purchaser and relating to the business of the Company which, if known to the Purchaser, might reasonably be expected to have a materially adverse effect on the Company's business or deter the Purchaser from entering into this Agreement or from completing the transactions contemplated hereby;

I. the corporate records of the Company are complete and accurate in all material respects and all corporate proceedings and actions reflected therein have been conducted or taken in compliance with all applicable laws and with the constating documents of the Company, and all transactions, reorganisations or other actions of the Company have been duly authorized by all necessary corporate action;

J. no other consents or approvals are required from any person whatsoever in connection with the transactions contemplated hereby nor are necessary on the part of the Company to authorise the execution and delivery of this Agreement or the consummation by the Company of the transactions contemplated hereby.

The Company agrees that the Purchaser is entitled to rely on the representations and warranties of the Company contained in this section 5 and the Company will indemnify and hold harmless the Purchaser for any loss, cost, expense, damage or any other liability the Purchaser may suffer as a result of a misrepresentation by the Company

6. General Provisions.

A. Choice of Law. This Agreement shall be governed by the internal substantive laws, but not the choice of law rules, of New York.

B. Integration. The Development Agreement and this Agreement, including all exhibits hereto, represent the entire agreement between the parties with respect to the issuance of the Shares to the Purchaser and supersedes and replaces any and all prior written or oral agreements regarding the subject matter of this Agreement including, but not limited to, any representations made during any interviews, relocation discussions or negotiations whether written or oral.

C. *Notices*. Any notice, demand, offer, request or other communication required or permitted to be given by either the Company or the Purchaser pursuant to the terms of this Agreement shall be in writing and shall be deemed effectively given the earlier of (i) when received, (ii) when delivered personally, (iii) one business day after being delivered by facsimile (with receipt of appropriate confirmation), (iv) one business day after being deposited with an overnight courier service or (v) four days after being deposited in the U.S. mail, First Class with postage prepaid and return receipt requested, and addressed to the parties at the addresses provided to the Company (which the Company agrees to disclose to the other parties upon request) or such other address as a party may request by notifying the other in writing.

D. *Successors*. Any successor to the Company (whether direct or indirect and whether by purchase, merger, consolidation, liquidation or otherwise) to all or substantially all of the Company's business and/or assets shall assume the obligations under this Agreement and agree expressly to perform the obligations under this Agreement in the same manner and to the same extent as the Company would be required to perform such obligations in the absence of a succession. For all purposes under this Agreement, the term "Company" shall include any successor to the Company's business and/or assets which executes and delivers the assumption agreement described in this section or which becomes bound by the terms of this Agreement by operation of law. Subject to the restrictions on transfer set forth in this Agreement, this Agreement shall be binding upon Purchaser and its executors, administrators, successors and assigns.

E. Assignment; Transfers. Except as set forth in this Agreement and subject to Section 3D above, this Agreement, and any and all rights, duties and obligations hereunder, shall not be assigned, transferred, delegated or sublicensed by the Purchaser without the prior written consent of the Company. Any attempt by the Purchaser without such consent to assign, transfer, delegate or sublicense any rights, duties or obligations that arise under this Agreement shall be void. Except as set forth in this Agreement, any transfers in violation of any restriction upon transfer contained in any section of this Agreement shall be void, unless such restriction is waived in accordance with the terms of this Agreement.

F. Amendment and Waiver. Any term of the Agreement may be amended and the observance of any term of the Agreement may be waived (either generally or in a particular instance and either retroactively or prospectively), only with the written consent of the Company and the Purchaser. Any amendment or waiver effected in accordance with this Section 5.F shall be binding upon each holder of any Shares at the time outstanding, each future holder of all such securities and the Company

G. Either party's failure to enforce any provision of this Agreement shall not in any way be construed as a waiver of any such provision, nor prevent that party from thereafter enforcing any other provision of this Agreement. The rights granted both parties hereunder are cumulative and shall not constitute a waiver of either party's right to assert any other legal remedy available to it.

H. Purchaser Investment Representations and Further Documents. Each Party agrees upon request to execute any further documents or instruments necessary or reasonably desirable to carry out the purposes or intent of this Agreement, including (but not limited to) the applicable exhibits and attachments to this Agreement.

I. Severability. Should any provision of this Agreement be found to be illegal or unenforceable, the other provisions shall nevertheless remain effective and shall remain enforceable to the greatest extent permitted by law.

J. *Rights as Stockholder*. Subject to the terms and conditions of this Agreement, Purchaser shall have all of the rights of a stockholder of the Company with respect to the Shares from and after the date that Purchaser delivers a fully executed copy of this Agreement (including the applicable exhibits and attachments to this Agreement) to the Company, and until such time as Purchaser disposes of the Shares in accordance with this Agreement. Upon such transfer, Purchaser shall have no further rights as a holder of the Shares so purchased except (in the case of a transfer to the Company) the right to receive payment for the Shares so purchased in accordance with the provisions of this Agreement, and Purchaser shall forthwith cause the certificate(s) evidencing the Shares so purchased to be surrendered to the Company for transfer or cancellation.

K. Adjustment for Stock Split. All references to the number of Shares in this Agreement shall be adjusted to reflect any stock split, stock dividend or other change in the Shares which may be made after the date of this Agreement.

L. Counterparts. This Agreement may be executed in one or more counterparts, each of which will be deemed an original, but all of which together will constitute one and the same agreement. Facsimile copies of signed signature pages shall be binding originals.

M. Reliance on Counsel and Advisors. The parties acknowledge that Wilson Sonsini Goodrich & Rosati, Professional Corporation, is representing only the Company in this transaction. The parties acknowledge that they have had the opportunity to review this Agreement, including all attachments hereto, and the transactions contemplated by this Agreement with their own legal counsel, tax advisors and other advisors. The parties are relying solely on their own counsel and advisors and not on any statements or representations of the Company or its agents for legal or other advice with respect to the transactions contemplated by this Agreement.

(signature page follows)

The parties represent that they have read this Agreement in its entirety, have had an opportunity to obtain the advice of counsel prior to executing this Agreement and fully understand this Agreement. The Purchaser agrees to notify the Company of any change in his or her address below.

NEOMED INSTITUTE

(Signed) Donald Olds

Signature

Donald Olds Print Name

President and Chief Executive Officer Print Title

Address:

7171 Frederick-Banting Saint-Laurent, Quebec, Canada H4S 1Z9

ARTELO BIOSCIENCES, INC.

(Signed) Gregory D. Gorgas Signature

Gregory D. Gorgas Print Name

President and Chief Executive Officer
Print Title

Artelo Biosciences, Inc. (Signature Page to Common Stock Purchase Agreement)

EXHIBIT A

INVESTMENT REPRESENTATION STATEMENT

PURCHASER COMPANY	:	NEOMED Institute Artelo Biosciences, Inc.
SECURITY AMOUNT DATE	:	Common Stock 120,000 shares January 2, 2018

In connection with the purchase of the above-listed shares, the undersigned Purchaser represents to the Company as follows:

1. The Company may rely on these representations. It understands that the Company's sale of the shares to it has not been registered under the Securities Act of 1933, as amended (the "Securities Act"), because the Company believes, relying in part on my representations in this document, that an exemption from such registration requirement is available for such sale. It understands that the availability of this exemption depends upon the representations it is making to the Company in this document being true and correct.

2. It is purchasing for investment. It is purchasing the shares solely for investment purposes, and not for further distribution. Its entire legal and beneficial ownership interest in the shares is being purchased and shall be held solely for its account. Subject to Section 3D it is not a party to, and does not presently intend to enter into, any contract or other arrangement with any other person or entity involving the resale, transfer, grant of participation with respect to or other distribution of any of the shares. Its investment intent is not limited to their present intention to hold the shares for the minimum capital gains period specified under any applicable tax law, for a deferred sale, for a specified increase or decrease in the market price of the shares, or for any other fixed period in the future.

3. It can protect their own interests. It can properly evaluate the merits and risks of an investment in the shares and can protect its own interests in this regard, whether by reason of its own business and financial expertise, the business and financial expertise of certain professional advisors unaffiliated with the Company with whom it has consulted, or its preexisting business or personal relationship with the Company or any of its officers, directors or controlling persons.

4. It is informed about the Company. It is sufficiently aware of the Company's business affairs and financial condition to reach an informed and knowledgeable decision to acquire the shares. It has had opportunity to discuss the plans, operations and financial condition of the Company with its officers, directors or controlling persons, and have received all information they deem appropriate for assessing the risk of an investment in the shares.

5. It recognizes the economic risk. It realizes that the purchase of the shares involves a high degree of risk, and that the Company's future prospects are uncertain. It is able to hold the shares indefinitely if required, and is able to bear the loss of their entire investment in the shares.

6. It knows that the shares are restricted securities. It understands that the shares are "restricted securities" in that the Company's sale of the shares to them has not been registered under the Securities Act in reliance upon an exemption for non-public offerings. In this regard, it also understands and agrees that:

A. It must hold the shares indefinitely, unless any subsequent proposed resale by them is registered under the Securities Act, or unless an exemption from registration is otherwise available (such as Rule 144);

B. the Company is under no obligation to register any subsequent proposed resale of the shares by it; and

C. the certificate evidencing the shares will be imprinted with a legend which prohibits the transfer of the shares unless such transfer is registration is not required in the opinion of counsel for the Company.

7. It is familiar with Rule 144. It is familiar with Rule 144 adopted under the Securities Act, which in some circumstances permits limited public resales of "restricted securities" like the shares acquired from an issuer in a non-public offering. It understands that its ability to sell the shares under Rule 144 in the future is uncertain, and may depend upon, among other things: (i) the availability of certain current public information about the Company; and (ii) the resale occurring more than a specified period after their purchase and full payment (within the meaning of Rule 144) for the shares; and (iii) the Company's former or current status as a "shell company".

8. It knows that Rule 144 may never be available. It understands that the requirements of Rule 144 may never be met, and that the shares may never be saleable under the rule. It further understands that at the time it wishes to sell the shares, there may be no public market for the Company's stock upon which to make such a sale, or the current public information requirements of Rule 144 may not be satisfied, either of which may preclude them from selling the shares under Rule 144 even if the relevant holding period had been satisfied.

9. It knows that it is subject to further restrictions on resale It understands that in the event Rule 144 is not available to it, any future proposed sale of any of the shares by it will not be possible without prior registration under the Securities Act, compliance with some other registration exemption (which may or may not be available), or each of the following: (i) their written notice to the Company containing detailed information regarding the proposed sale, (ii) their providing an opinion of its counsel to the effect that such sale will not require registration, and (iii) the Company notifying them in writing that its counsel concurs in such opinion or that the Company otherwise consents to the proposed sale or consent. It understands that neither the Company not is counsel is obligated to provide it with any such opinion. It understands that although Rule 144 is not exclusive, the Staff of the SEC has stated that persons proposing to sell private placement securities other than in a registered offering or pursuant to Rule 144 will have a substantial burden of proof in establishing that an exemption from registration is available for such offers or sales, and that such persons and their respective brokers who participate in such transactions do so at their own risk.

10. It knows that it may have tax liability due to the uncertain value of the shares It understands that the Internal Revenue Service may successfully assert that the value of the shares on the date of their purchase is substantially greater than the appraisal of the Board of Directors. It understands that any additional value ascribed to the shares by such an IRS determination will constitute ordinary income to it as of the purchase date, and that any additional taxes and interest due as a result will be their sole responsibility payable only by it, and that the Company need not and will not reimburse it for that tax liability. It understands that if such additional value represents more than 25% of their gross income for the year in which the value of the shares is taxable, the IRS will have 6 years from the due date for filing the return (or the actual filing date of the return if filed thereafter) within which to assess them the additional tax and interest due.

11. Principal Place of Business. The address of their principal place of business is set forth on the signature page below.

12. No "bad actor" disqualification events. Neither it nor any person that would be deemed a beneficial owner of the shares (in accordance with Rule 506(d) of the Securities Act) is subject to any of the "bad actor" disqualifications described in Rule 506(d)(1)(i) through (viii) under the Securities Act, except as set forth in Rule 506(d)(2)(ii) or (iii) or (d)(3) under the Securities Act and disclosed, reasonably in advance of the purchase or acquisition of the shares, in writing in reasonable detail to the Company.

By signing below, it acknowledges its agreement with each of the statements contained in this Investment Representation Statement as of the date first set forth above, and its intent for the Company to rely on such statements in issuing the shares to it.

NEOMED INSTITUTE

(signed) Donald Olds Purchaser's Signature

Donald Olds Print Name

President and Chief Executive Officer Print Title

Address: 7171 Frederick-Banting Saint-Laurent, Quebec, Canada H4S 1Z9

EXHIBIT E

COMMON STOCK PURCHASE AGREEMENT

ARTELO BIOSCIENCES, INC. COMMON STOCK PURCHASE AGREEMENT

This Common Stock Purchase Agreement (this "Agreement") is made as of [___], by and between Artelo Biosciences, Inc., a Nevada corporation (the "Company"), and NEOMED Institute, a not-for-profit corporation established under the *Not-for-Profit Corporations Act* (Canada) (the "Purchaser").

Reference is hereby made to the Material and Data Transfer, Option and License Agreement dated as of the date hereof by and between the Company and the Purchaser (the "**Development Agreement**"). Capitalized terms used in this Agreement that are not otherwise defined herein shall have the respective meanings assigned to them in the Development Agreement.

In consideration of the mutual covenants and representations set forth below, the Company and Purchaser agree as follows:

1. Issuance of the Shares. Subject to the terms and conditions of this Agreement, partial consideration for the license granted to the Company by Purchaser under Section 4.1 of the Development Agreement and pursuant to the terms of Section 5.2.1 of the Development Agreement, the Company hereby issues to the Purchaser on the Closing (as defined below) [___][1] shares of the Company's common stock (the "Shares").

2. Closing. The purchase and sale of the Shares shall occur at a closing (the 'Closing'') to be held on the date hereof.

3. Restrictions on Transfer.

A. Investment Representations and Legend Requirements. The Purchaser hereby makes the investment representations listed on Exhibit A to the Company as of the date of this Agreement, and agrees that such representations are incorporated into this Agreement by this reference, such that the Company may rely on them in issuing the Shares. Purchaser understands and agrees that the Company shall cause the legends set forth below, or substantially equivalent legends, to be placed upon any certificate(s) evidencing ownership of the Shares, together with any other legends that may be required by the Company or by applicable state or federal securities laws:

THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 (THE "*ACT*") AND MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED, PLEDGED OR HYPOTHECATED UNLESS AND UNTIL REGISTERED UNDER THE ACT OR, IN THE OPINION OF COUNSEL SATISFACTORY TO THE ISSUER OF THESE SECURITIES, SUCH OFFER, SALE OR TRANSFER, PLEDGE OR HYPOTHECATION OTHERWISE COMPLIES WITH THE ACT.

 $[\]overline{1}$ Equal to 2% of Artelo's Fully Diluted Shares. "**Fully-Diluted Shares**" shall mean, as of immediately prior to Artelo's exercise of the Option, the sum of (a) the outstanding shares of common stock of Artelo; (b) the shares of common stock of Artelo directly or indirectly issuable upon conversion or exchange of all outstanding securities directly or indirectly convertible into or exchangeable for common stock of Artelo and the exercise of all outstanding awards; and (c) the shares of common stock of Artelo reserved, but neither issued nor the subject of outstanding awards, under any equity incentive or similar plan of Artelo.

THE SHARES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO CERTAIN RESTRICTIONS ON TRANSFER IN THE EVENT OF A PUBLIC OFFERING AS SET FORTH IN THE COMMON STOCK PURCHASE AGREEMENT BETWEEN THE ISSUER AND THE ORIGINAL HOLDER OF THESE SHARES, A COPY OF WHICH MAY BE OBTAINED AT THE PRINCIPAL OFFICE OF THE ISSUER. SUCH TRANSFER RESTRICTIONS ARE BINDING ON TRANSFEREES OF THESE SHARES.

B. Stop-Transfer Notices. Purchaser agrees that to ensure compliance with the restrictions referred to herein, the Company may issue appropriate "stop transfer" instructions to its transfer agent, if any, and that, if the Company transfers its own securities, it may make appropriate notations to the same effect in its own records.

C. *Refusal to Transfer*. The Company shall not be required (i) to transfer on its books any Shares that have been sold or otherwise transferred in violation of any of the provisions of this Agreement or (ii) to treat as owner of such Shares or to accord the right to vote or pay dividends to any purchaser or other transferee to whom such Shares shall have been so transferred.

D. *Limitation on restriction:* Notwithstanding anything contained in this Agreement (including Exhibit A), the Company acknowledges and agrees that the Purchaser has the right to transfer [***] at any time provided that the terms and conditions of this Agreement (including Exhibit A) equally apply to [***].

4. *Tax Consequences.* The Purchaser has reviewed with the Purchaser's own tax advisors the federal, state, local and foreign tax consequences of this investment and the transactions contemplated by this Agreement. The Purchaser is relying solely on such advisors and not on any statements or representations of the Company or any of its agents. The Purchaser understands that the Purchaser (and not the Company) shall be responsible for any tax liability that may arise as a result of the transactions contemplated by this Agreement.

5. Representations and Warranties of the Company

The Company hereby makes the following representations and warranties to the Purchaser as of the date of this Agreement and acknowledges that the Purchaser is relying on such representations and warranties in entering into this Agreement:

A. the Company is a corporation duly formed and validly existing under the laws of Nevada and has all necessary corporate power and authority to execute this Agreement and any certificate evidencing ownership of the Shares (the "**Certificate**") and to offer, sell and issue the Shares, and to perform its obligations under this Agreement and the Certificate;

B. the authorized capital of the Company consists of 150,000,000 shares of common stock, of which 11,352,302 are currently issued and outstanding;

C. the Company has complied and will comply with all applicable corporate and securities laws and regulations in connection with the offer, sale and issuance of the Shares, and the execution, delivery and performance of this Agreement and the transactions contemplated by this Agreement will not result in any violation of or constitute either a default under any provision of its organizational documents or of any material agreement or instrument to which the Company is a party or by which the Company is bound, or any judgment, decree, order, law, statute, rule or regulation applicable to the Company;

D. this Agreement has been duly and validly authorized, executed and delivered by the Company and constitutes a legal, valid and binding obligation of the Company, enforceable against the Company in accordance with its terms;

E. the issuance of the Shares has been duly authorized and approved by all requisite corporate, regulatory and other action and upon Closing the Shares will be validly issued as fully paid and non-assessable, and shall be free and clear of any liens, mortgages, charges, security interests, burdens, encumbrances and other restrictions or limitations of any kind whatsoever;

F. the Company is duly qualified to transact business in each jurisdiction where the nature and extent of its business and properties (both owned and leased) requires;

G. the Company has been conducting its business in the ordinary course in compliance with all applicable laws, rules and regulations, including but not limited to federal, state, municipal and local laws, rules and regulations of each jurisdiction in which it carries on business and is not in breach of any such laws, rules or regulations;

H. the Company does not have any information or knowledge of any fact not communicated to the Purchaser and relating to the business of the Company which, if known to the Purchaser, might reasonably be expected to have a materially adverse effect on the Company's business or deter the Purchaser from entering into this Agreement or from completing the transactions contemplated hereby;

I. the corporate records of the Company are complete and accurate in all material respects and all corporate proceedings and actions reflected therein have been conducted or taken in compliance with all applicable laws and with the constating documents of the Company, and all transactions, reorganisations or other actions of the Company have been duly authorized by all necessary corporate action;

J. no other consents or approvals are required from any person whatsoever in connection with the transactions contemplated hereby nor are necessary on the part of the Company to authorise the execution and delivery of this Agreement or the consummation by the Company of the transactions contemplated hereby.

The Company agrees that the Purchaser is entitled to rely on the representations and warranties of the Company contained in this section 5 and the Company will indemnify and hold harmless the Purchaser for any loss, cost, expense, damage or any other liability the Purchaser may suffer as a result of a misrepresentation by the Company

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment

has been requested with respect to the omitted portions.

A. Choice of Law. This Agreement shall be governed by the internal substantive laws, but not the choice of law rules, of New York.

B. Integration. The Development Agreement and this Agreement, including all exhibits hereto, represent the entire agreement between the parties with respect to the issuance of the Shares to the Purchaser and supersedes and replaces any and all prior written or oral agreements regarding the subject matter of this Agreement including, but not limited to, any representations made during any interviews, relocation discussions or negotiations whether written or oral.

C. *Notices*. Any notice, demand, offer, request or other communication required or permitted to be given by either the Company or the Purchaser pursuant to the terms of this Agreement shall be in writing and shall be deemed effectively given the earlier of (i) when received, (ii) when delivered personally, (iii) one business day after being delivered by facsimile (with receipt of appropriate confirmation), (iv) one business day after being deposited with an overnight courier service or (v) four days after being deposited in the U.S. mail, First Class with postage prepaid and return receipt requested, and addressed to the parties at the addresses provided to the Company (which the Company agrees to disclose to the other parties upon request) or such other address as a party may request by notifying the other in writing.

D. Successors. Any successor to the Company (whether direct or indirect and whether by purchase, merger, consolidation, liquidation or otherwise) to all or substantially all of the Company's business and/or assets shall assume the obligations under this Agreement and agree expressly to perform the obligations under this Agreement in the same manner and to the same extent as the Company would be required to perform such obligations in the absence of a succession. For all purposes under this Agreement, the term "Company" shall include any successor to the Company's business and/or assets which executes and delivers the assumption agreement described in this section or which becomes bound by the terms of this Agreement by operation of law. Subject to the restrictions on transfer set forth in this Agreement, this Agreement shall be binding upon Purchaser and its heirs, executors, administrators, successors and assigns.

E. Assignment; Transfers. Except as set forth in this Agreement and subject to Section 3D above, this Agreement, and any and all rights, duties and obligations hereunder, shall not be assigned, transferred, delegated or sublicensed by the Purchaser without the prior written consent of the Company. Any attempt by the Purchaser without such consent to assign, transfer, delegate or sublicense any rights, duties or obligations that arise under this Agreement shall be void. Except as set forth in this Agreement, any transfers in violation of any restriction upon transfer contained in any section of this Agreement shall be void, unless such restriction is waived in accordance with the terms of this Agreement.

F. Amendment and Waiver. Any term of the Agreement may be amended and the observance of any term of the Agreement may be waived (either generally or in a particular instance and either retroactively or prospectively), only with the written consent of the Company and the Purchaser. Any amendment or waiver effected in accordance with this Section 5.F shall be binding upon each holder of any Shares at the time outstanding, each future holder of all such securities and the Company

G. Either party's failure to enforce any provision of this Agreement shall not in any way be construed as a waiver of any such provision, nor prevent that party from thereafter enforcing any other provision of this Agreement. The rights granted both parties hereunder are cumulative and shall not constitute a waiver of either party's right to assert any other legal remedy available to it.

H. Purchaser Investment Representations and Further Documents. Each Party agrees upon request to execute any further documents or instruments necessary or reasonably desirable to carry out the purposes or intent of this Agreement, including (but not limited to) the applicable exhibits and attachments to this Agreement.

I. Severability. Should any provision of this Agreement be found to be illegal or unenforceable, the other provisions shall nevertheless remain effective and shall remain enforceable to the greatest extent permitted by law.

J. *Rights as Stockholder*. Subject to the terms and conditions of this Agreement, Purchaser shall have all of the rights of a stockholder of the Company with respect to the Shares from and after the date that Purchaser delivers a fully executed copy of this Agreement (including the applicable exhibits and attachments to this Agreement) to the Company, and until such time as Purchaser disposes of the Shares in accordance with this Agreement. Upon such transfer, Purchaser shall have no further rights as a holder of the Shares so purchased except (in the case of a transfer to the Company) the right to receive payment for the Shares so purchased in accordance with the provisions of this Agreement, and Purchaser shall forthwith cause the certificate(s) evidencing the Shares so purchased to be surrendered to the Company for transfer or cancellation.

K. Adjustment for Stock Split. All references to the number of Shares in this Agreement shall be adjusted to reflect any stock split, stock dividend or other change in the Shares which may be made after the date of this Agreement.

L. Counterparts. This Agreement may be executed in one or more counterparts, each of which will be deemed an original, but all of which together will constitute one and the same agreement. Facsimile copies of signed signature pages shall be binding originals.

M. Reliance on Counsel and Advisors. The parties acknowledge that Wilson Sonsini Goodrich & Rosati, Professional Corporation, is representing only the Company in this transaction. The parties acknowledge that they have had the opportunity to review this Agreement, including all attachments hereto, and the transactions contemplated by this Agreement with their own legal counsel, tax advisors and other advisors. The parties are relying solely on their own counsel and advisors and not on any statements or representations of the Company or its agents for legal or other advice with respect to the transactions contemplated by this Agreement.

(signature page follows)

The parties represent that they have read this Agreement in its entirety, have had an opportunity to obtain the advice of counsel prior to executing this Agreement and fully understand this Agreement. The Purchaser agrees to notify the Company of any change in his or her address below.

NEOMED INSTITUTE

ARTELO BIOSCIENCES, INC.

Signature

Donald Olds Print Name

President and Chief Executive Officer Print Title

Address:

7171 Frederick-Banting Saint-Laurent, Quebec, Canada H4S 1Z9 Signature

Gregory D. Gorgas Print Name

President and Chief Executive Officer Print Title

Artelo Biosciences, Inc. (Signature Page to Common Stock Purchase Agreement)

EXHIBIT A

INVESTMENT REPRESENTATION STATEMENT

PURCHASER	:	NEOMED Institute
COMPANY	:	Artelo Biosciences, Inc.
RECUDITY		Common Storal
SECURITY	:	Common Stock
AMOUNT	:	[] shares
DATE :	:	[]

In connection with the purchase of the above-listed shares, the undersigned purchaser represents to the Company as follows:

1. The Company may rely on these representations. It understands that the Company's sale of the shares to it has not been registered under the Securities Act of 1933, as amended (the "Securities Act"), because the Company believes, relying in part on my representations in this document, that an exemption from such registration requirement is available for such sale. It understands that the availability of this exemption depends upon the representations it is making to the Company in this document being true and correct.

2. It is purchasing for investment. It is purchasing the shares solely for investment purposes, and not for further distribution. Its entire legal and beneficial ownership interest in the shares is being purchased and shall be held solely for its account. Subject to Section 3D it is not a party to, and do not presently intend to enter into, any contract or other arrangement with any other person or entity involving the resale, transfer, grant of participation with respect to or other distribution of any of the shares. Its investment intent is not limited to their present intention to hold the shares for the minimum capital gains period specified under any applicable tax law, for a deferred sale, for a specified increase or decrease in the market price of the shares, or for any other fixed period in the future.

3. It can protect their own interests. It can properly evaluate the merits and risks of an investment in the shares and can protect its own interests in this regard, whether by reason of its own business and financial expertise, the business and financial expertise of certain professional advisors unaffiliated with the Company with whom it has consulted, or its preexisting business or personal relationship with the Company or any of its officers, directors or controlling persons.

4. It is informed about the Company. It is sufficiently aware of the Company's business affairs and financial condition to reach an informed and knowledgeable decision to acquire the shares. It has had opportunity to discuss the plans, operations and financial condition of the Company with its officers, directors or controlling persons, and have received all information they deem appropriate for assessing the risk of an investment in the shares.

5. It recognizes the economic risk. It realizes that the purchase of the shares involves a high degree of risk, and that the Company's future prospects are uncertain. It is able to hold the shares indefinitely if required, and is able to bear the loss of their entire investment in the shares.

6. It knows that the shares are restricted securities. It understands that the shares are "restricted securities" in that the Company's sale of the shares to them has not been registered under the Securities Act in reliance upon an exemption for non-public offerings. In this regard, it also understands and agrees that:

A. It must hold the shares indefinitely, unless any subsequent proposed resale by them is registered under the Securities Act, or unless an exemption from registration is otherwise available (such as Rule 144);

B. the Company is under no obligation to register any subsequent proposed resale of the shares by it; and

C. the certificate evidencing the shares will be imprinted with a legend which prohibits the transfer of the shares unless such transfer is registered or such registration is not required in the opinion of counsel for the Company.

7. It is familiar with Rule 144. It is familiar with Rule 144 adopted under the Securities Act, which in some circumstances permits limited public resales of "restricted securities" like the shares acquired from an issuer in a non-public offering. It understands that its ability to sell the shares under Rule 144 in the future is uncertain, and may depend upon, among other things: (i) the availability of certain current public information about the Company; and (ii) the resale occurring more than a specified period after their purchase and full payment (within the meaning of Rule 144) for the shares; and (iii) the Company's former or current status as a "shell company".

8. It knows that Rule 144 may never be available. It understands that the requirements of Rule 144 may never be met, and that the shares may never be saleable under the rule. It further understands that at the time it wishes to sell the shares, there may be no public market for the Company's stock upon which to make such a sale, or the current public information requirements of Rule 144 may not be satisfied, either of which may preclude them from selling the shares under Rule 144 even if the relevant holding period had been satisfied.

9. It knows that it is subject to further restrictions on resale It understands that in the event Rule 144 is not available to it, any future proposed sale of any of the shares by it will not be possible without prior registration under the Securities Act, compliance with some other registration exemption (which may or may not be available), or each of the following: (i) their written notice to the Company containing detailed information regarding the proposed sale, (ii) their providing an opinion of its counsel to the effect that such sale will not require registration, and (iii) the Company notifying them in writing that its counsel concurs in such opinion or that the Company otherwise consents to the proposed sale or consent. It understands that neither the Company nor its counsel is obligated to provide them with any such opinion. They understand that although Rule 144 is not exclusive, the Staff of the SEC has stated that persons proposing to sell private placement securities other than in a registered offering or pursuant to Rule 144 will have a substantial burden of proof in establishing that an exemption from registration is available for such offers or sales, and that such persons and their respective brokers who participate in such transactions do so at their own risk.

10. It knows that it may have tax liability due to the uncertain value of the shares It understands that the Internal Revenue Service may successfully assert that the value of the shares on the date of their purchase is substantially greater than appraisal of the Board of Directors. It understands that any additional value ascribed to the shares by such an IRS determination will constitute ordinary income to it as of the purchase date, and that any additional taxes and interest due as a result will be their sole responsibility payable only by it, and that the Company need not and will not reimburse it for that tax liability. It understands that if such additional value represents more than 25% of their gross income for the year in which the value of the shares is taxable, the IRS will have 6 years from the due date for filing the return (or the actual filing date of the return if filed thereafter) within which to assess them the additional tax and interest due.

11. Principal Place of Business. The address of their principal place of business is set forth on the signature page below.

12. No "bad actor" disqualification events. Neither it nor any person that would be deemed a beneficial owner of the shares (in accordance with Rule 506(d) of the Securities Act) is subject to any of the "bad actor" disqualifications described in Rule 506(d)(1)(i) through (viii) under the Securities Act, except as set forth in Rule 506(d)(2)(ii) or (iii) or (d)(3) under the Securities Act and disclosed, reasonably in advance of the purchase or acquisition of the shares, in writing in reasonable detail to the Company.

By signing below, it acknowledges its agreement with each of the statements contained in this Investment Representation Statement as of the date first set forth above, and its intent for the Company to rely on such statements in issuing the shares to it.

NEOMED INSTITUTE

Purchaser's Signature

Donald Olds Print Name

President and Chief Executive Officer Print Title

Address: 7171 Frederick-Banting Saint-Laurent, Quebec, Canada H4S 1Z9

SCHEDULE 9.2.6

WARRANTY DISCLOSURE

- 1. With respect to priority right assignment of application US 60/3640,306: NEOMED has the following documents:
 - a) [***] (Assignment A); and
 - b) [***] (Assignment B).

CERTIFICATION PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Greg Gorgas, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Artelo Biosciences, Inc.;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. I am responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting;
- 5. I have disclosed, based on my most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: January 16, 2018

/s/ Greg Gorgas Greg Gorgas

President, Chief Executive Officer, Chief Financial Officer, Treasurer and Director (Principal Executive Officer, Principal Financial Officer and Principal Accounting Officer)

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

The undersigned, Greg Gorgas, hereby certifies, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) the quarterly report on Form 10-Q of Artelo Biosciences, Inc. for the period ended November 30, 2017 (the "Report") fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of Artelo Biosciences, Inc.

Dated: January 16, 2018

/s/ Greg Gorgas

Greg Gorgas President, Chief Executive Officer, Chief Financial Officer, Treasurer and Director (Principal Executive Officer, Principal Financial Officer and Principal Accounting Officer)

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signature that appears in typed form within the electronic version of this written statement required by Section 906, has been provided to Artelo Biosciences, Inc. and will be retained by Artelo Biosciences, Inc. and furnished to the Securities and Exchange Commission or its staff upon request.